ASSESSMENT OF ACCOUNTING INFORMATION, MARKET VALUE, COST OF EQUITY AND MODERATING EFFECT OF INTELLECTUAL CAPITAL

MOHAMMADGHORBAN MEHRI

UNIVERSITI TEKNOLOGI MALAYSIA

ASSESSMENT OF ACCOUNTING INFORMATION, MARKET VALUE, COST OF EQUITY AND MODERATING EFFECT OF INTELLECTUAL CAPITAL

MOHAMMADGHORBAN MEHRI

A thesis submitted in fulfilment of the requirements for the award of the degree of Doctor of Philosophy (Management)

Faculty of Management Universiti Teknologi Malaysia

MARCH 2013

This work is dedicated to my parents Zahra and Hossein who provided unconditional love and taught me how to soar on eagle's wings and to my siblings Abdolreza, Mariam, Gholamreza, Laila, Fatemeh and my shaheed brother Mohammadreza.

ACKNOWLEDGMENT

I would like to thank above all else, Allah Almighty, for blessing me and truly giving me the faith, courage, inspiration, and steadfastness to continue on this journey in my life. I realised that He is always with me. I would like to thank my supervisors, Assoc. Prof. Dr. Saudah Sofian and Dr. Siti Zaleha Abdul Rasid who have continuously mentored me throughout my entire Ph.D. program and have shared their experiences, wisdom, and knowledge as a academicians as well as scholars. They shaped my doctoral experience and provided me with the necessary skills to be successful in my academic career. Without their unwavering help and guidance, this journey would not have been possible. I would also like to thank Dr. Maisarah Mohamed Saat, Dr. Ahmad Jusoh, Dr. Mohamat Sabri Hassan, Dr. Azian Bin Madun, Dr. Rohaida Basiruddin for their guidance, support, advice, and helpful comments. My special thanks to Islamic Azad University of Iran, Ghorveh Branch, which funded my studies and approved my study leave. I would like to thank my parents, Zahra and Hossein, who have always encouraged me and have taught me to think big, to never give up, and to believe in myself. I dedicate this dissertation to them for providing love, understanding, and encouragement during the pursuit of this degree. To my siblings Mohammadreza, Abdolreza, Mariam, Gholamreza, Laila, Fatemeh, I thank them for their care, concern, and understanding. I am thankful to many other people who have contributed to my success in completing this final degree.

ABSTRACT

It is contended among the accounting researchers that information asymmetry could be reduced through disclosure of information pertaining to quality of earnings and recognition of intellectual capital. Correspondingly, this study examined the relationship between accounting earnings attributes i.e. relevance and conservatism of earnings and cost of equity. This study included an investigation of the moderating effect of intellectual capital on (i) the relationship between earnings and book value (individually and aggregate) and market price, and (ii) the relevance of earnings and cost of equity. The data for the study were based on annual financial reports, Datastream and Bloomberg of firms from the Technology, Trading and Services, Consumer Products and Hotel sectors listed in the main board of Bursa Malaysia from 2006 to 2010. The study employed multiple and moderated multiple regression to analyse data. The results showed that, firstly, firms with high quality of relevance of earnings benefit more from lower cost of equity and there is no relationship between conservatism of earnings and cost of equity, Secondly, the moderating of intellectual capital has an effect on the relationship between earnings and book value with market price. Lastly, the findings showed that intellectual capital has a moderating effect on relevance of earnings and cost of equity. This study contributes to the valuation theory through extension of Ohlson model by the inclusion of new-knowledge of intellectual capital. Besides that, it contributes to the understanding of signaling theory through the explanation of the moderating effect of intellectual capital on the relationship between relevance of earnings and cost of equity. The findings of the study has shown the importance of intellectual capital and it is recommended that accounting standard setting authorities must pave the avenue for the disclosure of intellectual capital in a firm's annual financial report.

ABSTRAK

Penyelidik dalam bidang perakaunan menegaskan bahawa tahap ketidakselarian maklumat dapat dikurangkan dengan adanya pendedahan maklumat tentang kualiti perolehan dan pengiktirafan modal intelek sebagai aset. Selari dengan itu kajian ini telah mengenal pasti hubungan antara sifat perolehan perakaunan, contohnya, kerelevanan dan konservatisme perolehan, dengan kos ekuiti. Kajian ini juga mengkaji kesan modal intelek sebagai moderator kepada (i) hubungan antara perolehan dan nilai buku (secara berasingan dan keseluruhan) dengan harga pasaran, dan (ii) hubungan antara kerelevanan perolehan dengan kos ekuiti. Data kajian ini diperoleh daripada penyata kewangan tahunan; Datastream dan Bloomberg bagi firma dalam sektor Teknologi, Perdagangan dan Perkhidmatan, Barangan Pengguna dan Hotel yang tersenarai di papan utama Bursa Malaysia dari tahun 2006 hingga 2010. Data dianalisis menggunakan multiple dan moderated multiple regression. Hasil kajian yang pertama menunjukkan bahawa firma dengan kualiti relevan yang tinggi memperoleh keuntungan yang lebih berbanding dengan kos ekuiti yang rendah dan tidak wujud hubungan antara konsertivisme dengan kos ekuiti. Kedua, modal intelek sebagai moderator memberikan kesan kepada hubungan antara perolehan dan nilai buku dengan harga pasaran. Akhir sekali, hasil kajian menunjukkan bahawa modal intelek mempunyai kesan sebagai moderator kepada kerelevanan perolehan dan kos ekuiti. Kajian ini menyumbang kepada valuation theory melalui lanjutan model Ohlson dengan menambah pengetahuan baru tentang modal intelek. Selain itu kajian ini menyumbang kepada signaling theory dengan memberikan penjelasan tentang kesan modal intelek sebagai moderator kepada hubungan antara kerelevanan perolehan dengan kos ekuiti. Kajian ini menunjukkan kepentingan modal intelek dan mencadangkan bahawa pihak berkuasa yang terlibat dalam piawaian perakaunan haruslah membuka satu saluran bagi pendedahan modal intelek dalam laporan kewangan tahunan firma.

CHAPTER 1

INTRODUCTION

1.1 Overview

The relationship between accounting information and capital markets have attracted a considerable level of attention and the contribution of Ball and Brown in 1968 believed to be a pioneering work in this regard. Up till now this is one of the most popular issues in the accounting literature. The interest for this subject is legitimate, given the generally accepted statement that accounting figures are aimed at providing investors with relevant information for their investment decisions. Diamond (1985), Diamond and Verrecchia (1991) and Embong et al. (2012) noted that high quality of accounting information disclosure reduces the information risk (information asymmetry) between market participants in financial markets and as a consequence, reduces the cost of equity. Lang and Lundholm (1996), Botosan (1997), Botosan and Plumlee (2002), and Easley et al. (2002) empirically examined and supported this conjecture. Watts and Zimmerman (1986) advocated the most significant reason for establishing regulations and standards for accounting information disclosure is reducing the information asymmetry between management and external users particularly investors. Financial Accounting Standard Board (FASB) No. 2 (1980) states that relevance and reliability are two primary determinants to assess the quality of accounting information and deemed useful for investors, which guide them in decision making process. Consequently, to attract large number of investors and concurrently minimizing the cost of capital, firms

generally disclose substantial sets of information especially those related to the earnings for investors to facilitate them in decision process.

Information pertaining to earnings is deemed significant by the accounting researchers in accounting literature as it guides the investor's decision process (Schipper and Vincent, 2003). Several accounting researches documented that poor quality of earnings lead to the increase in information asymmetry (risk) and consequently, higher cost of equity (Aboody *et al.*, 2005; Bhattacharya *et al.*, 2007; Francis *et al.*, 2004; Lara *et al.*, 2010; Ng, 2008; Verdi, 2008; Abdul-Latiff and Mohd-Taib, 2011). While on the other hand, reduction in information asymmetry results in decreased cost of equity. Firms that concern about quality of earnings disclose more information pertaining to earnings.

Recently, information concerning the earnings quality has become debatable as a result of several accounting scandals around the world like Worldcom and Enron. Similarly, several firms in Malaysia (Wimems Corp Bhd., Media Holdings Bhd. and Nasioncom Holdings Bhd.) falsified their accounting information in terms of receivables, tangible assets and revenues, and in results collapsed (Zaimee, 2007). These scandals lead to serious concerns about the quality and usefulness of reported earnings that will be used by users of financial information to evaluate their decisions. This requires on the part of the key players in local corporate governance such as Bursa Malaysia, Securities Commission, corporate advisors, auditors, managers and accountants to pave the avenue to prevent the accounting scandals to restore the confidence of the investors.

This discussion signifies the decisive role of earnings information in investment decisions reached by the investors in general. The dearth of literature on the topic pertaining to the Asian firms requires the extent of investigation in Asia to be further extended to assess the investment patterns adopted by the investors in Asia. This study assessed the effect of relevance / conservatism of earnings on cost of equity in Malaysia to provide quality information related to earning disclosed by firms to facilitate investors in the process of decision making, as quality and accuracy of information leads to the reduction in CoE, and the quality and quantity of information influence on CoE (Chan *et al.*, 2009; Francis *et al.*, 2005, 2004; Lara *et al.*, 2010; Pae *et al.*, 2005).

Many studies have been and continue to be conducted in several developed countries to observe the relationship between earnings attributes and Cost of Equity (Francis et al., 2004; Wong, 2008; Petruska, 2008; Chan et al., 2009; Pae et al., 2005; Lara et al., 2010; Li, 2010) still there is a need to re-examine this matter in developing countries in view of the fact that earning is recognized as significant accounting information for investors to guide them in making rationale and profitable decisions. To have a more clearer picture it can be said that, in an environment where information is already rich, any additional information disclosed by companies may not have much effect on decision-making process on investors and may not significantly affect the CoE. However, Malaysian market is a good platform to investigate disclosure issue where the legal system and capital market are well developed (Mohamad et al., 2007) but the accounting information environment is not rich (Ball et al., 2003). In addition, there is prevalent doubt in literature on the notion where conservatism of earnings leads to reduction of the CoE. Francis et al. (2004) reported that there is no association between conservatism of earnings and CoE. While, the findings reported by Lara et al. (2010), Li (2010), Petruska (2008) and Wong (2008) advocated that conservatism of earnings has a negative effect on CoE. Therefore, this study makes an attempt to enrich the conservatism of earnings literature, and investigated the relationship between conservatism of earnings and CoE.

Parallel to this debate, which stresses on the significant role of earnings information in guiding the decision environment, economic theorists assert that land, labour and money are three significant resources that contribute to the success of corporate enterprises (Sullivan, 1998). However, the last two decades witnessed the transition as the emphasis has shifted from traditional economic system to knowledge intensive system (Phillips and Phillips, 2002; Amidon, 2003). This change in emphasis created the hype for service industries all over the world. The dominance of service industries in terms of swiping the major share of value creation process has led the world to recognize intellectual capital as another decisive actor of knowledge-based economy, as it plays pivotal role in the firm overall growth. Intellectual capital in recent years has turned a major source of firm competitive advantage (Holland, 2003). Thus, growth patterns in knowledge intensive economy require re-evaluating the key drivers of growth, as traditional accounting practices

have failed to recognize in full the knowledge elements as assets of the firms in financial reporting system. Therefore, the non-recognition of intellectual capital resulted in increasing the gap between market and book values of firms (Amir and Lev, 1996; Brennan, 2001; Holland, 2003; Lev, 2001). Moreover, Generally Accepted Accounting Principles (GAAPs) and accounting standards, which discourage the recognition of many intangible assets, make financial information less relevant (Collins et al., 1997; Francis and Schipper, 1999; Lev and Zarowin, 1999). It is recently contended by the researchers that conservatism has an essential role in recognizing intellectual capital (intangible asset), and stands as a premier source of gap between market value and book value and also diminishes the relevance of accounting information over time (Beaver and Ryan, 2005). Furthermore, recognition of intellectual capital decreases the information asymmetries between companies and the users of financial reports (Byrnes and Derhovanesian, 2002; McNamee, 2001; Reed et al., 2002; Cormier et al., 2009; Wyatt and Frick, 2010) and consequently, diminishes the cost of equity (Mangena et al., 2010; Orens et al., 2009).

Aligned with the development of intellectual capital and the knowledge-based economy, Malaysian business regulatory agencies started to promote the development of knowledge-driven economy in the early 2000s. Knowledge Economy Master Plan, which was published in 2000, consisted of strategies, which aimed at transforming Malaysia from an input-driven to a knowledge-driven economy. As Mahathir Mohamad (former Prime Minister of Malaysia) in his keynote address during Global Knowledge Conference II, 2000 stated:

Vision 2020 emphasizes that in the information age which we have entered – our society must be information rich. There was a time when land was the most fundamental basis for prosperity and wealth, then came the second wave, the age of industrialisation. Now, increasingly knowledge will not only be the basis of power but also prosperity...Through the shift to k-economy, where the knowledge content and the knowledge contribution will see a quantum leap in every area, the Malaysian economy and Malaysia's society will not be quite the same again.

Although, Hassan and Mohd-Saleh (2010) revealed that earnings and book value were relevant during 1999, 2000, 2002 and 2003, in Malaysian context the disclosure of information still seems to be at low or minimal level (Ball *et al.*, 2003; Mohd Ghazali and Weetman, 2006). In addition, in regard with the role of intellectual capital in knowledge-based economy particularly in order to achieve the Malaysian's national mission (Master Plan was published in 2000), and also to increase the disclosure of information, this study contributed to the larger body of knowledge (Goodwin and Ahmed, 2006; Vafaei *et al.*, 2011) by determining: the impact of intellectual capital on the relationship between accounting earnings and book value (individually and aggregate) on market price.

Moreover, it is a general observation that many firms disclose considerable sets of information for some reasons like attracting a large number of investors in capital market particularly in Malaysian capital market since the information environment is not affluent (Ball *et al.*, 2003), another reason could be recognition of intellectual capital which leads to decrease the information asymmetries (Byrnes and Derhovanesian, 2002; McNamee, 2001; Reed *et al.*, 2002; Cormier *et al.*, 2009; Wyatt and Frick, 2010) and consequently, diminishes the cost of equity (Mangena *et al.*, 2010; Orens *et al.*, 2009). Hence this study examined: the effect of intellectual capital on the relationship between relevance of earnings and cost of equity. In line with the discussion, statement of problem is discussed in the subsequent section.

1.2 Statement of the Problem

Capital market structure contributed significantly to the development and growth of world economy, while on the other hand, investment stands as a significant indicator of economic development in capital market structure. This signifies that information based guided decisions in optimum capital structure plays a substantial role in reducing uncertainty and risk attached to investor's decision criteria (Diamond, 1985; Diamond and Verrecchia, 1991; Embong *et al.*, 2012).

Contemporary research contends that information asymmetry is reduced through disclosure of information by the firms in the capital market structure. Therefore, enhancement and improvement in the quality of information pertaining to earnings and recognition of intellectual capital lead to the reduction in information asymmetry (information risk), and concurrently increase the investor's willingness to invest, and consequently reduces cost of equity (Diamond and Verrecchia, 1991; Easley and O'Hara, 2004; Glosten and Milgrom, 1985). Aligned with this discussion, this study examined whether: (i) earnings quality attributes (relevance of earnings and conservatism of earnings) have an effect on cost of equity; (ii) intellectual capital and its components (human capital and structural capital) influence the relationship between earnings and book value with market price; and (iii) intellectual capital and its components (human capital and structural capital) influence the association between relevance of earnings and cost of equity.

Earnings is a significant accounting information and impacts largely on stock price behaviour and its profitability (Bhattacharya *et al.*, 2007; Dechow and Shrand, 2004; Financial Accounting Standard Board (FASB), 1980; Kim and Kross, 2005; Lipe, 1990; Penman, 2003; Richardson *et al.*, 2005; Schipper and Vincent, 2003; Warfield and Wild, 1992). Empirical researches illustrate that investors relied more on earnings' information than any other information including dividend earnings and cash flows (Bhattacharya *et al.*, 2007; Dechow and Shrand, 2004; Dimitropoulos *et al.*, 2010; Financial Accounting Standard Board (FASB), 1980; Penman, 2003; Schipper and Vincent, 2003). As noted by Francis *et al.* (2004) managers consider earnings as a key standard for investors' decision criteria.

Similarly, cost of capital is another fundamental concept of financial accounting literature. It is commonly used in corporate business models to assess the company valuation and shaping the corporate strategy. Cost of capital shapes the world by determining the parallel between investment, economic growth and consumption at both macroeconomic and microeconomic level (Ogier *et al.*, 2004). Discussions in literature suggest the significant role of earnings and cost of capital especially cost of equity in capital market to guide investors in their decision making process (Aboody *et al.*, 2005; Bhattacharya *et al.*, 2007; Francis *et al.*, 2004; Lara *et al.*, 2010; Ng, 2008; Verdi, 2008; Abdul-Latiff and Mohd-Taib, 2011). This signifies that the amount and preciseness of information influence cost of capital; as high

quality information causes more coordination among managers and investors in investment decisions. Contrary to that, if information asymmetry between managers and investors increases, investors claim more cost of capital because of the associated risk. Levitt (1997) stated that the quality of accounting standards improves liquidity and reduces the cost of capital and Financial Accounting Standard Board (FASB) in 2001 echoed similar sentiments. On the other hand literature demonstrates that poor quality of earnings lead to increase in information asymmetry and consequently higher cost of equity (Aboody *et al.*, 2005; Bhattacharya *et al.*, 2007; Francis *et al.*, 2004; Lara *et al.*, 2010; Ng, 2008; Verdi, 2008; Abdul-Latiff and Mohd-Taib, 2011).

This study accordingly posits that high and low level of cost of equity depend on the information's quality, which is related to earnings. Therefore, this study addresses the gaps in literature by extending the understanding on the relationship between earnings attributes and cost of equity, and aims to reduce the issue pertaining to information asymmetry, which complicates the decision environment for investors by hypothesizing the relationship between accounting earning attributes (relevance of earnings and conservatism of earnings) and cost of equity. Relevance of earnings and conservatism of earnings were the selected attributes of earnings in this study because of following reasons;

- The most important usefulness reason of accounting system is providing relevant information, which is standing as a utility guide for investors in decision making process (FASB No. 2, 1980). In addition, earnings is deemed significant by the accounting researchers in accounting literature as it guides the investor's decision process (Schipper and Vincent, 2003). However, the earnings literature posits that the relevance of earnings has diminished (Collins *et al.*, 1997; Francis and Schipper, 1999; Lev and Zarowin, 1999). Therefore, based on the important role of relevance of earnings among accounting information for investors in decision making, this study selected relevance of earnings to examine whether the relevance of earnings is increased or decreased and also investigated whether recognition of intellectual capital can effect on the relevance of earnings.
- Accounting conservatism literature illustrates two types of conservatism: (1) ex post conservatism (earnings conservatism); and (2) ex ante conservatism (balance

sheet conservatism) (Beaver and Ryan, 2005; Chan *et al.*, 2009). The ex post conservatism is associated to more timely recognition of bad news than of good news while ex ante conservatism develops from the application of Generally Accepted Accounting Principles (GAAPs) and accounting standards, which reduces the book value of net assets (primarily due to unrecognized intangible assets). Since this study claimed that ex ante conservatism has essential role in the recognition of IC (intangible asset), and it generates gap between market value and book value and that it diminishes the relevance of earnings over time, therefore, to clarify the differences and better understanding of these two types of conservatism this study chose to examine conservatism of earnings as one of attributes of earnings.

Analogous to the above arguments, financial accounting literature illustrates that contemporary competitive business environment requires organizations to explore new patterns of knowledge to acquire competitive edge over competitors. Unfortunately, traditional accounting practices fail to accord new patterns of knowledge in financial reporting and consequently limit the access of accurate information to users (Byrnes and Derhovanesian, 2002; McNamee, 2001; Reed et al., 2002; Cormier et al., 2009; Wyatt and Frick, 2010). This issue has been debated over and over again that non-recognition of new-knowledge (intellectual capital) causes a gap between market value and book value of the company (Amir and Lev, 1996; Brennan, 2001; Holland, 2003). In addition to that it is concurrently argued that the value relevance of traditional financial information has decreased because conventional financial reporting system merely presents an underestimated company value (Collins et al., 1997; Francis and Schipper, 1999; Lev and Zarowin, 1999), where, the estimation of true market value requires re-evaluation of the key intellectual factors on the part of the firms, which eventually contribute to the overall growth of the firm.

Non-recognition of intellectual capital increases information asymmetry between firms and external users of financial reports (Byrnes and Derhovanesian, 2002; McNamee, 2001; Reed *et al.*, 2002; Cormier *et al.*, 2009; Wyatt and Frick, 2010). Empirical studies on this subject area have pointed to the problem pertaining to the lack in reporting of knowledge based information in financial reports and

proposed new patterns to overcome the deficiencies of traditional financial reporting systems by disclosing ample accounting information (Blair and Wallman, 2000; Bose and Oh, 2004; Gelb, 2002; Lev, 2001). Brooking (1996) argued that intellectual capital represents 62% of overall firm value. To pave the avenue to encourage firms and recognize the significance of intellectual capital in financial reporting for the purpose of decreasing the information asymmetry in capital market, it is timely and significant for accounting researchers to resolve this issue through exemplified empirical research. Correspondingly, this study evaluated the effect of intellectual capital and its components (human capital and structural capital) on the relationship between earnings and book value (individually and aggregate) and market price. This study also examined the effect of intellectual capital and its components (human capital and structural capital) on relevance of earnings and cost of equity to address this longstanding gap in accounting research.

In relation to the above, this study employed value added intellectual capital for measuring intellectual capital which categorizes in three components: (i) human capital efficiency; (ii) structural capital efficiency; and (iii) capital employed efficiency. On the other hand, this study utilized price based model (Ohlson, 1995) to assess the value relevance of accounting information which consists of two main components: book value (the net value of investment) and earnings (the present value of the period benefits). Therefore, to examine the effect of intellectual capital on the relationship between earnings and book value (individually and aggregate) and market price, this study ignored capital employed as a component of value added intellectual capital since it was included in price based model.

Cultivating from above discussion, this study determined:

- Whether there is any relationship between earnings attributes (relevance of earnings and conservatism of earnings) and cost of equity.
- Whether intellectual capital and its components (human capital and structural capital) influence the relationship between earnings and book value (individually and aggregate) and market price.

 Whether intellectual capital and its components (human capital and structural capital) influence the association between relevance of earnings and cost of equity.

1.3 Purpose of the Study

The aim of this study was to assess the relationships between; Accounting earnings quality attributes (relevance of earnings and conservatism of earnings) and cost of equity; Role of intellectual capital and its components (human capital and structural capital) in connection to its effect as moderator on the relationship between earnings and book value (individually and aggregate) and market price; and The effect of intellectual capital and its components (human capital and structural capital) on relationship between relevance of earnings and cost of equity.

1.4 Objectives of the Study

The research problem examined in this study was designed with the objectives listed below:

- To assess whether earning attributes (relevance of earnings and conservatism of earnings) have a significant and negative effect on cost of equity.
- To examine whether intellectual capital and its components (human capital and structural capital) as moderators have an effect on the relationship between earnings and book value (individually and aggregate) and market price.

 To determine whether the intellectual capital and its components (human capital and structural capital) have an effect on the relationship between relevance of earnings and cost of equity.

1.5 Significance of the Study

This study contributes to the body of knowledge by responding to the gaps in accounting literature through assessing the association between earnings attributes (relevance of earnings and conservatism of earnings) and cost of equity in Malaysia since to the best of the author's knowledge, no research has reported the association between earnings attributes (relevance of earnings and conservatism of earnings) and CoE in the context of Malaysian firms. This study is considered significant in several ways: first, the issue pertaining to the reduction in quality of accounting information is frequently highlighted in literature (Brown et al., 1999; Chang, 1998; Collins et al., 1997; Francis and Schipper, 1999; Lev and Zarowin, 1999). However, lack of empirical research based on intellectual capital and accounting information makes it decisive to expand the horizon of research and seek answers to the questions that relate to the direct and indirect impact of intellectual capital on accounting information (Ahmed and Falk, 2006; Reinita, 2007). Second, this study contributes potentially by suggesting to the accounting standard setting authorities to modify the traditional accounting standards and conventional accounting practices to standardize the stipulations of modern and competitive business era, and pave the avenue for the recognition of IC as an asset to the firm and motivating researchers to expand the horizon of research and seek solution to the underlying questions (non-recognition of intellectual capital). Lastly, this study is a pioneer study, which assesses the effect of intellectual capital on the relevance of earnings and cost of equity, unlike previous studies which emphasised more on the examination of the effect of intellectual capital on firm performance only (Chan, 2009; F-Jardo'n and Martos, 2009; Gan and Saleh, 2008; Ghosh and Mondal, 2009; Goh, 2005; Ismail, 2005; Kamath, 2007;

Marimuthu et al., 2009; Sofian, 2005; Ting and Lean, 2009; Ze'ghal and Maaloul, 2010).

1.6 Scope of the Study

This study being empirical, evaluated the relationship between accounting earning attributes (relevance of earnings and conservatism of earnings) and cost of equity. Subsequently, it examined the moderation effect of intellectual capital and its components (human capital and structural capital) on the relationship between accounting earnings and book value (individually and aggregate) on market price, and finally, determined the moderation effect of intellectual capital and its components (human capital and structural capital) on the relationship between relevance of earnings and cost of equity. In line with the study objectives, this research was carried out in high intangible intensive listed companies of Bursa Malaysia (Technology industry, Trading and Services industry, Consumer Products industry and Hotel industry) covering the period from 2006 to 2010. The data needed for this study were obtained from 1999 to 2010, to construct firm-year observations from 2006 to 2010.

1.7 Outline of the Thesis

This research consists of seven chapters. Chapter 1 discusses the research problem, aims and objectives of the study, Chapter 2 reviews the literature on the relationship between earnings attributes and cost of equity. This chapter also explicates the evidences related to earning attributes, quality of earnings and cost of equity. Chapter 3 discusses intellectual capital in connection to how and why companies intend to recognize intellectual capital as an asset and measurement of

valuation models of intellectual capital. Chapter 4 delineates the research framework and hypotheses, while Chapter 5 discusses the research methodology in connection to the testing of the study hypotheses. Chapter 6 presents the results and discusses the findings of the study, and Chapter 7 presents the conclusion of the study after discussing the findings and contribution of the study. This is followed by description and limitations of the study and possible avenues for further research.

1.8 Terminologies

Following specific terminologies use in this study and define as bellow:

- Conservatism of Balance Sheet / Ex ante / Unconditional Conservatism:
 Ex ante conservatism stems from the application of Generally Accepted Accounting Principles (GAAPs) and accounting standards, this reflects reduced book value of net assets (primarily due to unrecognized intangible assets) (Pae et al. 2005).
- Conservative of Earning / Ex post / Conditional Conservatism: Ex post conservatism is also called conditional conservatism, marked-based, and news-dependent conservatism and is associated to more timely recognition of bad news than of good news. The effect of this kind of conservatism on earnings streams is less predictable and persistent (Pae et al., 2005).
- Cost of Equity: Shareholders have a claim on the value of the company when firms issue shares and this claim has been repaid after debt. Therefore, shareholders receive two benefits from the firm i.e. dividend and increase in the value of shares (Ross *et al.*, 2010).
- Earnings Quality: Financial Accounting Standard Board (FASB) No. 2 (1980) defined high quality earnings as "representational faithfulness" that refers to the correspondence or agreement between a

- measure/description and the phenomenon it purports to represent (para.63).
- Human Capital: Represent the skills, experiences, and firm' know-how at the collective level, and also at the individual level, capacity of employees, suppliers to solve the problems of customer (Sullivan, 1999).
- Human Capital Efficiency: The value which is created by one unit of investment in the employees is called human capital efficiency. Human capital efficiency is measured from value added divided by human capital (Pulic, 1998).
- Information Asymmetry: Information asymmetry is the gap information between manager and investors. This signifies that when information asymmetry between managers and investors increases, investors claim more cost of capital because of the associated risk (Diamond and Verrecchia, 1991).
- Intellectual Capital: Malaysian Financial Reporting Standards No. 138
 defines intangible asset or intellectual capital as, "An intangible asset is
 an identifiable non-monetary asset without physical substance"
 (Malaysian Financial Reporting Standards No. 138, para. 8).
- Intellectual Capital Efficiency: It shows how efficiently intellectual capital has created value. Intellectual capital efficiency (ICE) is measured by adding up the human capital efficiency and structural capital efficiency (Pulic, 1998).
- Relevance of Earning: According to Financial Accounting Standard Board (FASB) No. 2 (1980) "relevance and reliability are two primary criteria to improve the quality of accounting information", which help stakeholders in decision process. This defines relevance as "accounting information capable of making a difference in a decision" (para. 51). For the information to be relevant, it should have: (a) predictive or feedback value; and (2) presented timely.

- Structural Capital: Structure capital is direct support and indirect support
 of the human capital. Direct support which touch the human capital such
 as computer, computer software and information system. Indirect
 support includes electricity, payroll system that does not touch the human
 capital directly (Sullivan, 1999).
- Structural Capital Efficiency: Structural capital efficiency shows the value added efficiency of structural capital. Structural capital efficiency is measured from structural capital divided by value added (Pulic, 2004).

1.9 Summary

This chapter has discussed the issues raised in the study, its significance and comprised of three main objectives: first, this study examined the relation between earnings attributes (relevance of earnings and conservatism of earnings) and cost of equity; second, this study investigated the effect of intellectual capital and its components (human capital and structural capital) on the relationship between accounting earnings and book value (individually and aggregate) and market price; lastly, this study assessed the effect of intellectual capital and its components (human capital and structural capital) on relevance of earnings and cost of equity. Staying in line with the study objectives, this study was carried out in high intangible intensive listed companies of Bursa Malaysia (Technology industry, Trading and Services industry, Consumer Products industry and Hotel industry) covering the period from 2006 to 2010.

REFERENCES

- Abayadeera, N. (2010). Value Relevance of Information in High-Tech Industries in Australia: Financial and Non-Financial. PhD Thesis, Victoria University, Melbourne, Australia.
- Abeysekera, I. (2011). The Relation of Intellectual Capital Disclosure Strategies and Market Value in Two Political Settings. *Journal of Intellectual Capital*, 12(2), 319-338.
- Aboody, D., Hughes, J. and Liu, J. (2005). Earnings Quality, Insider Trading, and Cost of Capital. *Journal of Accounting Research*, 43(5), 651-673.
- Aboody, D. and Lev, B. (1998). The Value Relevance of Intangibles: The Case of Software Capitalization. *Journal of Accounting Research*, 36(3), 161-191.
- Aboody, D. and Lev, B. (2000). Information Asymmetry, R&D, and Insider Gains. *The Journal of Finance*, 55(6), 2747-2766.
- Ahmed, K. and Falk, H. (2006). The Value Relevance of Management's Research and Development Reporting Choice: Evidence from Australia. *Journal of Accounting and Public Policy*, 25(3), 231-264.
- AICPA. (1994). Improving Business Reporting-A Customer Focus: Meeting the Information Needs of Investors and Creditors, Comprehensive Report of the Special Committee on Financial Reporting. New York, NY: AICPA.
- Aiken, L. S., West, S. G. and Reno, R. R. (1991). *Multiple Regression: Testing and Interpreting Interactions*. Newbury Park: CA: Sage.
- Akhtaruddin, M. (2005). Corporate Mandatory Disclosure Practices in Bangladesh. *The International Journal of Accounting*, 40(2005), 399–422.

- Aleksanyan, M. (2009). Does the Information Environment Affect the Value Relevance of Financial Statement Data? *Applied Economics Letters*, 16(8), 835–839.
- Alfaraih, M. and Alanezi, F. (2011). The Usefulness of Earnings and Book Value for Equity Valuation to Kuwait Stock Exchange Participants. *International Business & Economics Research Journal*, 10(1), 73-90.
- Allee, V. (2000). The Value Evolution: Addressing Larger Implications of an Intellectual Capital and Intangibles Perspective. *Journal of Intellectual Capital*, 1(1), 17-32.
- Amidon, D. M. (2003). The Innovation Superhighway: Harnessing Intellectual Capital for Sustainable Collaborative Advantage. United States of America: Butterworth Heinemann.
- Amir, E. and Lev, B. (1996). Value-Relevance of Nonfinancial Information: The Wireless Communications Industry. *Journal of Accounting and Economics*, 22(1-3), 3-30.
- Anandarajan, A. and Hasan, I. (2010). Value Relevance of Earnings: Evidence from Middle Eastern and North African Countries. *Advances in Accounting, Incorporating Advances in International Accounting*, 26(2), 270-279.
- Andersen, R. and McLean, R. (2000). Accounting for the Creation of Value.

 Ongoing Research Project Sponsored by the Canadian Institute of Chartered Accountants.
- Andriessen, D. (2004). Making Sense of Intellectual Capital: Designing the Method for the Valuation of Intangibles. Burlington, USA: Elsevier Butterworth-Heinman.
- Andriessen, D. and Tissen, R. (2000). Weightless Weight-Find Your Real Value in a Future of Intangible Assets. London: Pearson Education.
- Arvidsson, S. (2011). Disclosure of Non-Financial Information in the Annual Report:

 A Management-Team Perspective. *Journal of Intellectual Capital*, 12(2), 277-300.
- Asl, H. J. (2008). *Capital Structure and Cost of Capital: A Case of Malaysia*. Master Thesis, Multimedia University, Putrajaya-Cyberjaya.

- Bae, K.-H. and Jeong, S. W. (2005). The Value-Relevance of Earnings and Book Value, Ownership Structure, and Business Group Affiliation: Evidence From Korean Business Groups. *Journal of Business Finance & Accounting*, 34(5&6), 740-766.
- Balachandran, S. and Mohanram, P. (2011). Is the Decline in the Value Relevance of Accounting Driven by Increased Conservatism? *Review of Accounting Studies*, 16(2), 272-301.
- Ball, R. and Brown, P. (1968). An Empirical Evaluation of Accounting Income Numbers. *Journal of Accounting Research*, 6(2), 159-178.
- Ball, R., Robin, A. and Wu, J. S. (2003). Incentives Versus Standards: Properties of Accounting Income in Four East Asian countries. *Journal of Accounting and Economics*, 36(1/3), 235–270.
- Bao, B.-H. and Bao, D.-H. (2004). Income Smoothing, Earnings Quality and Firm Valuation. *Journal of Business Finance & Accounting*, 31(9&10), 1525-1555.
- Baron, R. M. and Kenny, D. A. (1986). The Moderator-Mediator Variable Distinction in Social Psychological Research: Conceptual, Strategic, and Statistical Considerations. *Pernality and Social Psychology*, 51(6), 1173-1182.
- Barsky, N., Catanach, A., Rhoades-Catanach, S. and Thibodeau, J. (2003). Enron: Did the Financial Reporting Model Really Fail? *Commercial Lending Review*, 18(2), 5-9.
- Barth, M. E., Beaver, W. H. and Landsman, W. R. (1998). Relative Valuation Roles of Equity Book Value and Net Income as a Function of Financial Health. *Journal of Accounting and Economics*, 25(1), 1-34.
- Barth, M. E., Cram, D. R. and Nelson, K. K. (2001). Accruals and the Prediction of Future Cash Flows. *The Accounting Review*, 76(1), 27-58.
- Barth, M. E., Konchitchki, Y. and Landsman, W. R. (2010). Cost of Capital and Earnings Transparency. *Available at:* http://ssrn.com/abstract=1348245 (accessed 30 September 2012).
- Barua, A. (2006). *Using the FASB's Qualitative Characteristics in Earnings Quality Measures*. PhD Thesis, Louisiana State University, Louisiana.

- Basu, S. (1997). The Conservatism Principle and The Asymmetric Timeliness of Earnings. *Accounting & Economics*, 24(1), 3-37.
- Beattie, V. and Thomson, S. J. (2004). A Comprehensive Analysis of Intellectual Capital Categories as a Precursor to Empirical Investigation of Disclosures in Annual Reports. *Paper presented at the Annual Financial Reporting and Business Communication Conference*, Cardiff, Wales.
- Beattie, V. and Thomson, S. J. (2007). Lifting the Lid on the Use of Content Analysis to Investigate Intellectual Capital Disclosures. *Accounting Forum*, 31(2), 129-163.
- Beaver, W. H. (1989). Financial Reporting: an Accounting Revolution (3rd ed.). Englewood Cliffs, N.J: Prentice Hall.
- Beaver, W. H. and Ryan, S. G. (2005). Conditional and Unconditional Conservatism: Concepts and Modeling. *Review of Accounting Studies*, 10(2), 269-309.
- Berenson, M. L., Levine, D. M. and Krehbiel, T. C. (2005). *Basic Business Statistics:* Concepts and Applications (10th ed.). Chicago: Prentice Hall.
- Bhattacharya, N., Desai, H. and Venkataraman, K. (2007). Earnings Quality and Information Asymmetry: Evidence from Trading Costs. *Working Paper*. *Southern Methodist University*.
- Bisman, J. (2010). Postpositivism and Accounting Research: A (Personal) Primer on Critical Realism. *Australasian Accounting Business and Finance Journal*, 4(4), 1-25.
- Blackburn, S. (1996). *Oxford Dictionary of Philosophy*. Oxford: Oxford University Press.
- Blair, M. M. and Wallman, S. M. H. (2000). *Unseen Wealth*. Washington, D.C: The Brookings Institution.
- Blanchard, O. J. (1987). Comment. *Journal of Business and Economic Statistics*, 5(4), 449-451.
- Boedker, C., Guthrie, J. and Cuganesan, S. (1993). The Strategic Significance of Human Capital Information in Annual Reporting. *Journal of Human Resource Costing & Accounting*, 8(2), 7-22.

- Bontis, N. (1996). The Strategic Management of Intellectual Capital & Organisational Knowledge. New York: Oxford University Press, Inc.
- Bontis, N. (2001). Assessing Knowledge Assets: A Review of the Models Used to Measure Intellectual Capital. *International Journal of Management Reviews*, 3(1), 41-60.
- Bontis, N. and Girardi, J. (2000). Teaching Knowledge Management and Intellectual Capital Lessons: An Empirical Examination of the Tango Simulation. *International Journal of Technology Management*, 20(5-8), 545-555.
- Bose, S. and Oh, K. B. (2004). Measuring Strategic Value-Drivers for Managing Intellectual Capital. *An International Journal*, 11(4/5), 347-356.
- Botosan, C. A. (1997). Disclosure Level and the Cost of Equity Capital. *The Accounting Review*, 72(3), 323-349.
- Botosan, C. A. (2006). Disclosure and the Cost of Equity Capital: What Do We Know? *Accounting and Business Research*, 36(1), 31-40.
- Botosan, C. A. and Plumlee, M. (2000). Disclosure Level and Expected Cost of Equity Capital: An Examination of Analysts' Rankings of Corporate Disclosure and Alternative Methods of Estimating Expected Cost of Equity Capital. *Working Paper. University of Utah*.
- Botosan, C. A. and Plumlee, M. A. (2002). A Re-Examination of Disclosure Level and the Expected Cost of Equity Capital. *Journal of Accounting Research*, 40(1), 21-40.
- Botosan, C. A. and Plumlee, M. A. (2005). Assessing Alternative Proxies for the Expected Risk Premium. *The Accounting Review*, 8(1), 21-53.
- Brennan, N. (2001). Reporting Intellectual Capital in Annual Reports: Evidence from Ireland. *Accounting, Auditing & Accountability Journal*, 14(4), 423-436.
- Brennan, N. and Connell, B. (2000). Intellectual Capital: Current Issues and Policy Implications. *Journal of Intellectual Capital*, 1(3), 206-240.
- Breusch, T. S. and Pagan, A. R. (1979). A Simple Test for Heteroscedasticity and Random Coefficient Variation. *Econometrica*, 47(5), 1287-1294.

- Brimble, M. and Hodgson, A. (2007). On the Intertemporal Value Relevance of Conventional Financial Accounting in Australia. *Accounting and Finance*, 47(4), 599-622.
- Brooking, A. (1996). *Intellectual Capital: Core Assets for the Third Millenium Enterprise*. London: International Thomson Business Press.
- Brown, S., Lo, K. and Lys, T. (1999). Use of r² in Accounting Research: Measuring Changes in Value Relevance over the Last Four Decades. *Journal of Accounting Research*, 28(2), 83–115.
- Bublitz, B. and Ettredge, M. (1989). The Information in Discretionary Outlays: Advertising, Research, and Development. *Accounting Review*, 64(1), 108-124.
- Bujang, I. and Nassir, A. M. (2007). The Relevance of Gordon's Model and Earnings Multiplier Approaches in Emerging Stock Market: Test with Appropriate Refinements. *Finance and Economics*, 2007(7), 141-152.
- Bukh, P. N. (2003). The Relevance of Intellectual Capital Disclosure: a Paradox? *Accounting, Auditing & Accountability Journal*, 16(1), 49-56.
- Bursa Malaysia Stock Exchange. (2010). Companies Listed in Stock Exchange.

 Available at: www.bursamalaysia.com (accessed 14 Appril 2011).
- Busenitz, L. W., Fiet, J. O. and Moesel, D. D. (2005). Signaling in Venture Capitalist-New Venture Team Funding Decisions: Does It Indicate Long-Term Venture Outcomes? *Entrepreneurship Theory and Practice*, 29(1), 1-12.
- Byrnes, N. and Derhovanesian, M. (2002). Earnings: A Cleaner Look. *Business Week*, 27 May, 32-35.
- Certo, S. T. (2003). Influencing Initial Public Offering Investors with Prestige: Signaling with Board Structures. Academy of Management Review, 28(3), 432-446.
- Chan, A. L.-C., Lin, S. W. J. and Strong, N. (2009). Accounting Conservatism and the Cost of Equity Capital: UK Evidence. *Managerial Finance*, 35(4), 325-345.

- Chan, K. H. (2009). Impact of Intellectual Capital on Organisational Performance:

 An Empirical Study of Companies in the Hang Seng Index (Part 1). *The Learning Organization*, 16(1), 4-21.
- Chang, J. (1998). The Decline in Value Relevance of Earnings and Book Values.

 Working Paper. University of Pennsylvania. Pennsylvania.
- Chen, M.-C., Cheng, S.-J. and Hwang, Y. (2005). An Empirical Investigation of the Relationship between Intellectual Capital and Firms' Market Value and Financial Performance. *Journal of Intellectual Capital*, 6(2), 159-176.
- Cheng, C.-H. (2006).Information Asymmetry and Earning Industry. Available Management in Taiwanese Tech at: http://www.knjc.edu.tw/admin/aa/publish/pic-1/book/8/5%E9%84%AD%E5%AE%B6%E8%88%88.pdf (accessed 30 September 2012).
- Cho, D. (1988). The Impact of Risk Management Decisions on Firm Value: Gordons Growth Model Approach. *The Journal of Risk and Insurance*, 55(1), 118-131.
- Claesson, G. (2011). Which Model Gives Me the Most Accurate Share Price, the Dividend Discount Model or the Free Cash Flow to Equity model? Master Thesis, Jonkoping University.
- Cohen, D. A. (2008). Does Information Risk Really Matter? An Analysis of the Determinants and Economic Consequences of Financial Reporting Quality. Asia-Pacific Journal of Accounting and Economics, 15(2), 69-90.
- Cohen, J. (1978). Partial Products Are Interactions; Partialed Powers Are Curve Components. *Psychological Bulletin*, 85(4), 858–866.
- Cohen, J., Cohen, P., West, S. G. and Aiken, L. S. (2003). *Applied Multiple Regression/Correlation Analysis for the Behavioral Sciences* (3rd ed.). Mahwah, NJ: Erlbaum.
- Coller, M. and Yohn, T. L. (1997). Management Forecasts and Information Asymmetry: An Examination of Bid-Ask Spreads. *Journal of Accounting Research*, 35(2), 181-191.
- Collins, D. W. and Hribar, P. (2002). Errors in Estimating Accruals: Implications for Empirical Research. *Journal of Accounting Research*, 40(1), 105-134.

- Collins, D. W., Maydew, E. L. and Weiss, I. S. (1997). Changes in the Value-Relevance of Earnings and Book Values over the Past Forty Years. *Journal of Accounting & Economics*, 24(1), 39-67.
- Collins, D. W., Pincus, M. and Xie, H. (1999). Equity Valuation and Nagative Earnings: The Role of Book Value of Equity. *The Accounting Review*, 74(1), 29-61.
- Connelly, B. L., Certo, S. T., Ireland, R. D. and Reutzel, C. R. (2011). Signaling Theory: A Review and Assessment. *Journal of Management*, 37(1), 39-67.
- Cooper, D. J. and Sherer, M. J. (1984). The Value of Corporate Accounting Reports:

 Arguments for a Political Economy of Accounting. *Accounting, Organizations and Society*, 9(3-4), 207-232.
- Cormier, D., Aerts, W., Ledoux, M.-J. and Magnan, M. (2009). Attributes of Social and Human Capital Disclosure and Information Asymmetry between Managers and Investors. *Canadian Journal of Administrative Sciences*, 26(2009), 71-88.
- Creswell, J. W. (2003). *Research Design:Quantitative, Qualitative & Mixed Method Approaches* (2nd ed.). United States: SAGE Publications.
- Damodaran, A. (2006). Applied Corporate Finance (2nd ed.): John Wiley & Sons.
- Damodaran, A. (2012). *Investment Valuation: Tools and Techniques for Determining* the Value of Any Asset (3rd ed.). New Jersey: John Wiley & Sons.
- Dechow, P. M. and Dichev, I. D. (2002). The Quality of Accruals and Earnings: The Role of Accrual Estimation Errors. *The Accounting Review*, 77(4), 35-59.
- Dechow, P. M., Hutton, A. P. and Sloan, R. G. (1999). An Empirical Assessment of the Residual Income Valuation Model 1. *Journal of Accounting and Economics*, 26(1-3), 1-34.
- Dechow, P. M. and Shrand, C. M. (2004). *Earning Quality* (1st ed.). United States of America: Research Foundation of CFA Institute.
- DeFond, M. L. (2010). Earnings Quality Research: Advances, Challenges and Future Research. *Journal of Accounting and Economics*, 50(2/3), 402–409.

- Diamond, D. W. (1985). Optimal Releases of Information by Firm. *Journal of Finance*, 40(4), 1071-1094.
- Diamond, D. W. and Verrecchia, R. E. (1991). Disclosure, Liquidity and the Cost of Capital. *Journal of Finance*, 46(4), 1325-1359.
- Dimitropoulos, P. E., Asteriou, D. and Koumanakos, E. (2010). The Relevance of Earnings and Cash Flows in a Heavily Regulated Industry: Evidence from the Greek Banking Sector. *Advances in Accounting, Incorporating Advances in International Accounting*, 26(2), 290-303.
- Drucker, P. F. (1999). Knowledge-Worker Productivity: The Biggest Challenge. *California Management Review*, 41(2), 79-94.
- Drukker, D. M. (2003). Testing for Serial Correlation in Linear Panel-Data Models. *The Stata Journal*, 3(2), 168-177.
- Easley, D., Hvidkjaer, S. and O'hara, M. (2002). Is Information Risk a Determinant of Asset Returns? *Journal of Finance*, 57(5), 2185-2221.
- Easley, D. and O'Hara, M. (2004). Information and the Cost of Capital. *The Journal of Finance*, 59(4), 1553-1583.
- Easton, P. D. (2004). PE Ratios, PEG Ratios, and Estimating the Implied Expected Rate of Return on Equity Capital. *Accounting Review*, 79(1), 73-79.
- Easton, P. D. and Harris, T. S. (1991). Earnings as an Explanatory Variable for Returns. *Journal of Accounting Research*, 29(1), 19-36.
- Easton, P. D. and Monahan, S. J. (2005). An Evaluation of Accounting-Based Measures of Expected Returns. *The Accounting Review*, 80(2), 501-538.
- Eccles, R. G. and Mavrinac, S. C. (1995). Improving the Corporate Disclosure Process. *Sloan Management Review*, 36(4), 11-25.
- Eckel, N. (1981). The Income Smoothing Hypothesis Revisited. *A Journal of Accounting, Finance and Business Studies*, 17(1), 28-40.
- Edvinsson, L. and Malone, M. S. (1997). *Intellectual Capital: Realizing Your Company's True Value by Finding Its Hidden Brainpower*. New York: Harper Business.

- Edwards, E. O. and Bell, P. W. (1961). *The Theory and Measurement of Business Income*. Berkeley and Los Angeles: University of California Press.
- El-Gazzar, S. M., Finn, P. M. and Tang, C. (2009). The Value Relevance of Earnings and Nonearnings Information in Regulated and Deregulated Markets: The Case of the Airline Industry. *International Advances in Economic Research*, 15(88), 88-101.
- Embong, Z., Mohd-Saleh, N. and Hassan, M. S. (2012). Firm Size, Disclosure and Cost of Equity Capital. *Asian Review of Accounting*, 20(2), 119-139.
- Eng, L. L., Nabar, S. and Chng, C. K. (2005). The Predictive Value of Earnings, Cash flows and Accruals in the Period Surrounding the Asian Financial Crisis: Evidence from Hong Kong, Malaysia, Singapore and Thailand. International Financial Management and Accounting, 16(3), 165-193.
- F-Jardo'n, C. M. and Martos, M. S. (2009). Intellectual Capital and Performance in Wood Industries of Argentina. *Journal of Intellectual Capital*, 10(4), 600-616.
- Fama, E. F. and French, K. R. (1993). Common Risk Factors in the Returns of Stocks and Bonds. *Financial Economics*, 33(1), 3-56.
- Farooq, S. U., Ullah, S., Alam, W. and Shah, A. A. (2010). The Performance of Equity Valuation Models for High and Low Intangible Companies-A Case of United States. *European Journal of Economics, Finance and Administrative Sciences*, (20), 141-160.
- FASB. (1974). SFAS No. 2: Accounting for Research and Development Costs. Stamford, CT: FASB.
- FASB. (1985). SFAS No. 86: Accounting for the Costs of Computer Software to be Sold, Leased, or Otherwise Marketed. Stamford, CT: FASB.
- Financial Accounting Standard Board (FASB). (1978). Statement of Financial Accounting Concepts No. 1. Stamfort, CT: FASB.
- Financial Accounting Standard Board (FASB). (1980). Statement of Financial Accounting Concepts No. 2: Qualitative Characteristics of Accounting Information. Stamfort, CT: FASB.

- Fitz-Enz, J. (1994). *How to Measure Human Resource Management*. New York: McGraw-Hill.
- Flamholtz, E. G. and Main, E. D. (1999). Current Issues, Recent Advancements, and Future Direction in Human Resource Accounting. *Journal of Human Resource Costing and Accounting*, 4(1), 11-20.
- Foerster, S. R. and Sapp, S. G. (2005). The Dividend Discount Model in the Long-Run: A Clinical Study. *Journal of Applied Finance*, 15(2), 55-75.
- Forcese, D. and Richer, S. (1973). *Social Research Methods*. Englewood Cliffs, New Jersey: Prentice Hall.
- Fortune. (1984). June 25 (pp. 58-61).
- Francis, J., LaFond, R., Olsson, P. and Schipper, K. (2005). The Market Pricing of Accruals Quality. *Journal of Accounting and Economics*, 39(2), 295-327.
- Francis, J., Lafond, R., Olsson, P. M. and Schipper, K. (2004). Cost of Equity and Earnings Attributes. *The Accounting Review*, 79(4), 967-1010.
- Francis, J., Nanda, D. and Olsson, P. (2008). Voluntary Disclosure, Earnings Quality, and Cost of Capital. *Journal of Accounting Research*, 46(1), 53-99.
- Francis, J. and Schipper, K. (1999). Have Financial Statements Lost Their Relevance? *Journal of Accounting Research*, 37(2), 319-352.
- Frazier, P. A., Tix, A. P. and Barron, K. E. (2004). Testing Moderator and Mediator Effects in Counseling Psychology Research. *Journal of Counseling Technology*, 51(1), 115-134.
- French, C. W. (2002). Jack Treynor's 'Toward a Theory of Market Value of Risky Assets'. *Available at:*http://papers.ssrn.com/sol3/papers.cfm?abstract_id=628187 (accessed 30 September 2012).
- Fridson, M. and Alvarez, F. (2002). *Finanacial Statement Analysis* (3rd ed.). New York: John Wiley &Sons, Inc.
- Froidevaux, P. S. (2004). Fundamental Equity Valuation Stock Selection Based on Discounted Cash Flow. PhD Thesis, University of Fribourg, Switzerland.

- Froot, K. A. (1989). Consistent Covariance Matrix Estimation with Cross-Sectional Dependence and Heteroskedasticity in Financial Data. *Journal of Financial and Quantitative Analysis*, 24(3), 333-355.
- Gan, K. and Saleh, Z. (2008). Intellectual Capital and Corporate Performance of Technology-Intensive Companies: Malaysia Evidence. *Asian Journal of Business and Accounting*, 1(1), 113-130.
- Garcı´a-Ayuso, M. (2003). Intangibles Lessons from the Past and a Look into the Future. *Journal of Intellectual Capital*, 4(4), 597-604.
- Gebhardt, W. R., Lee, C. M. C. and Swaminathan, B. (2001). Toward and Implied Cost of Capital. *Accounting Research*, 39(1), 135-176.
- Gelb, D. S. (2002). Intangible Assets and Firms' Disclosures: An Empirical Investigation. *Journal of Business, Finance and Accounting*, 29(2&3), 457-476.
- Ghosh, S. and Mondal, A. (2009). Indian Software and Pharmaceutical Sector IC and Financial Performance. *Journal of Intellectual Capital*, 10(3), 369-388.
- Gillett, D. (2002). Spiritual Capital: Building Vibrant Business that Serve Shareholders & Humanity. United States: Institute of Human Economics.
- Givoly, D. and Hayn, C. (2000). The Changing Time-Series Properties of Earnings, Cash Flow and Accruals: Has Financial Reporting Become More Conservative? *Accounting & Economics*, 29(3), 287-320.
- Gjerde, Ø., Knivsflå, K. H. and Sættem, F. (2005). The Value Relevance of Financial Reporting on the Oslo Stock Exchange Over the Period 1964-2003.

 Discussion Papers 2005/23, Department of Finance and Management Science, Norwegian School of Economics and Business Administration.
- Glosten, L. R. and Milgrom, P. R. (1985). Bid, Ask, and Transaction Prices in a Specialist Market with Heterogeneously Informed Traders. *Journal of Financial Economics*, 14(1), 71-100.
- Gode, D. and Mohanram, P. (2003). Inferring the Cost of Capital Using the Ohlson-Juettner Model. *Review of Accounting Studies*, 8(4), 399-431.

- Goh, P. C. (2005). Intellectual Capital Performance of Commercial Banks in Malaysia. *Intellectual Capital*, 6(3), 385-396.
- Goldfinger, C. (1997). Understanding and Measuring the Intangible Economy: Current Status and Suggestions for Future Research. *Paper presented at the CIRET Seminar*, Helsinki.
- Goodwin, J. and Ahmed, K. (2006). Longitudinal Value Relevance of Earnings and Intangible Assets: Evidence from Australian Firms. *International Accounting, Auditing and Taxation*, 15(1), 72-91.
- Gordon, M. J. (1962). The Investment, Financing, and Valuation of the Corporation. *American Economic Review*, 52(5), 1174-1176.
- Graham, J. R., Harvey, C. R. and Rajgopal, S. (2005). The Economic Implications of Corporate Financial Reporting. *Journal of Accounting and Economics*, 40(1-3), 3-73.
- Gravetter, F. J. and Wallnau, L. B. (2009). *Statistics for the Behavioral Sciences* (8th ed.). United States of America: Wadsworth, Cengage Learning.
- Gray, P., Koh, P.-S. and Tong, Y. H. (2009). Accruals Quality, Information Risk and Cost of Capital: Evidence from Australia. *Journal of Business Finance & Accounting*, 36(1), 51-72.
- Greene, W. H. (2008). Econometric Analysis (6th ed.). USA: Pearson Education Inc.
- Groeneveld, R. A. and Meeden, G. (1984). Measuring Skewness and Kurtosis. *The Statistician*, 33(4), 391-399.
- Gu, F. and Lev, B. (2004). The Information Content of Royalty Income. *Accounting Horizons*, 18(1), 1-12.
- Gujarati, D. N. and Porter, D. C. (2008). *Basic Econometrics* (5th ed.). New York, USA: McGraw-Hill/Irwin.
- Guthrie, J., Petty, R. and Ricceri, F. (2006). The Voluntary Reporting of Intellectual Capital: Comparing Evidence from Hong Kong and Australia. *Journal of Intellectual Capital*, 7(2), 254-271.

- Guthrie, J., Petty, R. and Ricceri, F. (2007). *Intellectual Capital Reporting: Lessons from Hong Kong and Australia*. Edinburgh, Scotland: The Institute of Chartered Accountants of Scotland.
- Haanes, K. and Lowendahl, B. (1997). *The Unit of Activity: Towards an Alternative* to the Theories of the Firm, Structure and Style. New York: John Wiley & Sons, Inc.
- Hagerman, R. L. and Zmijewski, M. E. (1979). Some Economic Determinants of Accounting Choice. *Journal of Accounting & Economics*, 1(2), 141-161.
- Hair, J. F., Black, W. C., Babin, B. J. and Anderson, R. E. (2010). *Multivariate Data Analysis: A Global Perspective* (7th ed.). Upper Saddle River, New Jersey: Pearson Prentice Hall.
- Hair, J. F., Black, W. C., Babin, B. J., Anderson, R. E. and Tatham, R. L. (2006). *Multivariate Data Analysis* (6th ed.). USA: Pearson Prentice Hall.
- Hamilton, L. C. (2009). *Statistics with Stata: Update for Version 10*. USA: Brooks/Cole, Cengage Learning.
- Hassan, M. S. and Mohd-Saleh, N. (2010). The Value Relevance of Financial Instruments Disclosure in Malaysian Firms Listed in the Main Board of Bursa Malaysia. *Internation Journal of Economics and Management*, 4(2), 243-270.
- Hayton, J. C. (2005). Competing in the New Economy: the Effect of Intellectual Capital on Corporate Entrepreneurship in High-Technology New Ventures. *R&D Management*, 35(2), 137-155.
- Healy, P. M. (1984). The Effect of Bonus Schemes on Accounting Decisions. *Journal of Accounting and Economics*, 7(1-3), 85-107.
- Healy, P. M. and Wahlen, J. M. (1999). A Review of the Earnings Management Literature and Its Implications for Standard Setting. *Accounting Horizons*, 13(4), 365-383.
- Hellström, K. (2006). The Value Relevance of Financial Accounting Information in a Transition Economy: The Case of the Czech Republic. *European Accounting Review*, 15(3), 325–349.
- Hendriksen, E. (1992). Accounting Theory. Burr Ridge, IL: Irwin.

- Hoechle, D. (2007). Robust Standard Errors for Panel Regressions with Cross-Sectional Dependence. *The Stata Journal*, 7(3), 281-312.
- Holland, J. (2003). Intellectual Capital and the Capital Market-Organisation and Competence. *Accounting, Auditing & Accountability Journal*, 16(1), 39-48.
- Holland, J. (2004). *Corporate Intangibles, Value Relevance and Disclosure Content*. Edinburgh, Scotland: Institute of Chartered Accountants of Scotland.
- Holland, J. (2006). Fund Management, Intellectual Capital, Intangibles and Private Disclosure. *Managerial Finance*, 32(4), 277-316.
- Howell, D. C. (2011). Fundamental Statistics for the Behavioral Sciences (7th ed.). USA: Linda Schreiber.
- Hudson, W. J. (1993). *Intellectual Capital: How to Build It, Enhance It, Use It*. New York: John Wiley & Sons, Inc.
- ICAEW. (2003). Information for Markets: New Reporting Models for Business. London: Institute of Chartered Accountants in England and Wales.
- International Federation of Accountants (IFAC). (1998). The Measurement and Management of Intellectual Capital. London: IFAC.
- Ismail, M. B. (2005). The Influence of Intellectual Capital on the Performance of Telekom Malaysia. PhD Thesis, Universiti Teknologi Malaysia, Johor Bahru, Malaysia.
- Ittner, C., Kalafut, P., Larcker, D., Sean Love, S., Low, J., Park, J., et al. (2000). Introducing the New Value Creation Index. *Forbes ASAP. from www. forbes.com (accessed 30 September 2012)*.
- Jermakowicz, E. K., Prather-Kinsey, J. and Wulf, I. (2007). The Value Relevance of Accounting Income Reported by DAX-30 German Companies. *Journal of International Financial Management and Accounting*, 18(3), 151-191.
- Johanson, U. (1996). Increasing the Transparency of Investments in Intangibles: Personnel Economics Institute, School of Business, Stockholm University.
- Jurczak, J. (2008). Intellectual Capital Measurement Methods. *Economics and Organization of Enterprise*, 1(1), 37-45.

- Kadri, M. H., Aziz, R. A. and Ibrahim, M. K. (2009). Value Relevance of Book Value and Earnings: Evidence from Two Different Financial Reporting Regimes. *Journal of Financial Reporting & Accounting*, 7(1), 1-16.
- Kadri, M. H., Aziz, R. A. and Ibrahim, M. K. (2010). Balanced Scorecard Equity Valuation Model. *Paper presented at the International Conference on Science and Social Research*, Kuala Lumpur, Malaysia.
- Kamath, G. B. (2007). The Intellectual Capital Performance of the Indian Banking Sector. *Journal of Intellectual Capital*, 8(1), 96-123.
- Kaplan, R. S. and Norton, D. P. (1996). *The Balanced Scorecard: Translating Strategy into Action*. United States of America: Harvard Business School.
- Kasztler, A. and Leitner, K.-H. (2002). Bibliometric Analysis and Visualisation of Intellectual Capital. *Journal of Universal Computer Science*, 8(5), 516-525.
- Khan, S. (2011). The Interactive Effects of Intellectual Capital Components on the Relevance of the Balance Sheet as an Indicator of Corporate Value. *The Journal of American Academy of Business, Cambridge*, 16(2), 130-136.
- Kim, M. and Kross, W. (2005). The Ability of Earnings to Predict Future Operating Cash Flows Has Been Increasing-Not Decreasing. *Journal of Accounting Research*, 43(5), 753-780.
- Kim, O. and Verrecchia, R. E. (1994). Market Liquidity and Volume Around Earning Announcements. *Journal of Accounting and Economics*, 17(1), 41-67.
- Kincaid, H. (1998). Positivism in the Social Science, Routledge Encyclopedia of Philosophy. London: Routledge.
- Kirmani, A. and Rao, A. R. (2000). No Pain, No Gain: A Critical Review of the Literature on Signaling Unobservable Product Quality. *Journal of Marketing*, 64(2), 66-79.
- Koenker, R. and Gilbert Bassett, J. (1978). Regression Quantiles. *Econometrica*, 46(1), 33-50.
- Kothari, S. P. and Zimmerman, J. L. (1995). Price and Return Models. *Accounting and Economics*, 20(2), 155-192.

- Kristandl, G. and Bontis, N. (2007). The Impact of Voluntary Disclosure on Cost of Equity Capital Estimates in a Temporal Setting. *Journal of Intellectual Capital*, 8(4), 577-594.
- Kroll, C. N. and Stedinger, J. R. (1998). Regional Hydrologic Analysis: Ordinary and Generalized Least Squares Revisited. Water Resources Research, 34(1), 121-128.
- Kujansivu, P. and Lo"nnqvist, A. (2007). Investigating the Value and Efficiency of Intellectual Capital. *Journal of Intellectual Capital*, 8(2), 272-287.
- Kwon, G.-J. (2009). The Value Relevance of Book Values, Earnings and Cash Flows: Evidence from Korea. *International Journal of Business and Management*, 4(10), 28-42.
- Lai, F.-W., Azizan, N. A. and Samad, M. F. A. (2009). Shareholders Value Creation Through Enterprise Risk Management. *International Journal of Business Research*, 10(1), 44-57.
- Laing, G., Dunn, J. and Hughes-Lucas, S. (2010). Applying the VAIC Model to Australian Hotels. *Journal of Intellectual Capital*, 11(3), 269-283.
- Lang, M. H. and Lundholm, R. J. (1996). Corporate Disclosure Policy and Analyst Behavior. *The Accounting Review*, 71(4), 467-492.
- Lara, J. M. G., Osma, B. G. and Penalva, F. (2010). Conditional Conservatism and Cost of Capital. *Review of Accounting Studies*, 16(2), 247-271.
- Abdul-Latiff, R. A. and Mohd-Taib, F. M. (2011). The Cost of Equity Effects of Accruals Quality and Ownership Structure. *Academy of Taiwan Business Management Review*, 7(1), 1-10.
- Leadbeater, C. (2000). New Measures for the New Economy. London: Institute of Chartered Accountants in England and Wales.
- Lee, W.-S., Tzeng, G.-H., Guan, J.-L., Chien, K.-T. and Huang, J.-M. (2009). Combined MCDM Techniques for Exploring Stock Selection Based on Gordon Model. *Expert Systems with Applications*, 36(2009), 6421-6430.

- Lee, Y. M. and Whiting, R. H. (2011). Technology, Intellectual Capital Disclosure and Cost of Capital. *Paper presented at the AFAANZ Conference*, Darwin, Australia.
- Legatum Prosperity IndexTM. (2012). Legatum Prosperity Index. Available at: http://www.sec.gov/news/speech/speecharchive/1997/spch176.txt (accessed 30 September 2012).
- Lester, R. H., Certo, S. T., Dalton, C. M., Dalton, D. R. and Cannella, A. A. (2006). Initial Public Offering Investor Valuations: An Examination of Top Management Team Prestige and Environmental Uncertainty. *Journal of Small Business Management*, 44(1), 1-26.
- Leuz, C. and Verrecchia, R. E. (2000). The Economic Consequences of Increased Disclosure. *Journal of Accounting Research*, 38(3), 91-124.
- Lev, B. (1999). The Inadequate Public Information on Intellectual Capital and Its Consequences. *Paper presented at the Measuring and Reporting Intellectual Capital: Experience, Issues, and Prospects*, OECD Symposium on Intellectual Capital, Amsterdam.
- Lev, B. (2001). *Intangibles: Management, Measurement, and Reporting*. Washington, DC: The Brooking Institution.
- Lev, B. and Sougiannis, T. (1996). The Capitalization, Amortization, and Value-Relevance of R&D. *Journal of Accounting and Economics*, 21(1), 107-138.
- Lev, B. and Zambon, S. (2003). Introduction to the Special Issue. *European Accounting Review*, 12(4), 597-603.
- Lev, B. and Zarowin, P. (1999). The Boundaries of Financial Reporting and How to Extend Them? *Journal of Accounting Research*, 37(2), 353-385.
- Levitt, A. (1997). The Importance of High Quality Accounting Standards. US Securities and Exchange Commission to the Inter-American Development Bank in Washington, D.C. on September 29, 1997. Available at: http://www.sec.gov/news/speech/speecharchive/1997/spch176.txt (accessed 30 September 2012).

- Li, J., Pike, R. and Haniffa, R. (2008). Intellectual Capital Disclosure and Corporate Governance Structure in UK Firms. *Accounting and Business Research*, 38(2), 137-159.
- Li, X. (2010). Accounting Conservatism and The Cost of Capital: An International Analysis. Working Paper. Temple University Fox School of Business and Management.
- Lin, P. (2008). How Does R&D Capability Affect Market Value Under Different Life Cycle Stages? Available at:http://ir.lib.ntust.edu.tw:8080/dspace/handle/987654321/13753 (accessed 30 September 2012).
- Lintner, J. (1965). The Valuation of Risk Assets and the Selection of Risky Investments in Stock Portfolios and Capital Budgets. *Review of Economics and Statistics*, 47(1), 13-37.
- Lipe, R. (1990). The Relation Between Stock Returns and Accounting Earnings Given Alternative Information. *The Accounting Review*, 65(1), 49-71.
- Liu, J., Nissim, D. and Thomas, J. (2002). Equity Valuation Using Multiples. *Journal of Accounting Research*, 40(1), 135-172.
- Lubberink, M. J. P. (2000). *Financial Statment Information: The Impact of Investors and Managers*. PhD Thesis, Groningen, Groningen.
- Luthy, D. H. (1998). Intellectual Capital and Its Measurement. *Paper presented at the Asian Pacific Interdisciplinary Research in Accounting Conference (APIRA)*, Osaka, Japan.
- Mahmud, R., Ibrahim, M. K. and Pok, W. C. (2009). Earnings Quality Attributes and Performance of Malaysian Public Listed Firms. *Available at SSRN:* http://ssrn.com/abstract=1460309 or http://dx.doi.org/10.2139/ssrn.1460309 (accessed 30 September 2012).
- Malaysian Financial Reporting Standards No. 138. (2006). Intangible Assets.

 Malaysia: Malaysiam Accounting Standards Board.
- Mangena, M., Pike, R. and Li, J. (2010). *Intellectual Capital Disclosure Practices* and Effects on the Cost of Equity Capital: UK Evidence (1st ed.). Great Britain: T. J. International Ltd.

- Marimuthu, M., Arokiasamy, L. and Ismail, M. (2009). Human Capital Development and Its Impact on Firm Performance: Evidence from Developmental Economics. *The Journal of International Social Research*, 2(8), 265-272.
- Marr, B. and Schiuma, G. (2001). *Measuring and Managing Intellectual Capital and Knowledge Assets in New Economy Organisations*. London: Gee.
- Marr, B., Schiuma, G. and Neely, A. (2004). Intellectual Capital-Defining Key Performance Indicators for Organizational Knowledge Assets. *Business Process Management Journal*, 10(5), 551-569.
- Martens, J. (2008). *Reporting on Intellectual Capital*. Master Thesis, Universiteit Gent, Gent.
- Mayo, A. (2000). The Role of Employee Development in the Growth of Intellectual Capital. *Personnel Review*, 29(4), 521-533.
- McElroy, M. W. (2002). Social Innovation Capital. *Journal of Intellectual Capital*, 3(1), 30-39.
- McInnis, J. M. (2008). Are Smoother Earnings Associated with a Lower Cost of Equity Capital? PhD Thesis, University of Iowa, Iowa.
- McNamee, M. (2001). The Big Five Need to Factor in Investors. *Business Week*, 24 *December*, 34.
- McNichols, M. and Trueman, B. (1994). Public Disclosure, Private Information Collection and Short-Term Trading. *Journal of Accounting and Economics*, 17(1-2), 69-94.
- Meer-Kooistra, J. v. d. and Zijlstra, S. M. (2001). Reporting on Intellectual Capital. *Accounting, Auditing & Accountability Journal*, 14(4), 456-476.
- Meritum Project. (2002). Guidelines for Managing and Reporting on Intangibles.

 Madrid, Spain: Fundacion Aitel Movil.
- MFRS. (2006). see Malaysian Financial Reporting Standards.
- Miller, M. H. and Modiglini, F. (1961). Dividend Policy, Growth and Valuation of Shares. *The Journal of Business*, 34(4), 411-433.

- Modigliani, F. and Miller, M. H. (1958). The Cost of Capital, Corporation Finance, and the Theory of Investment. *The American Economic Review*, 48(3), 261-297.
- Modigliani, F. and Miller, M. H. (1963). Corporate Income Taxes and the Cost of Capital: A Correction. *The American Economic Review*, 53(3), 433-443.
- Mohamad, M. (2000). Building Knowledge Societies: Access- Empowerment-Governance. 2nd Global Knowledge Conference. Kuala Lumpur.
- Mohamad, S., Hassan, T. and Ariff, M. (2007). Research in an Emerging Malaysian Capital Market: A Guide to Future Direction. *International Journal of Economics and Management*, 1(2), 173-202.
- Mohd Ghazali, N. A. and Weetman, P. (2006). Perpetuating Traditional Influences: Voluntary Disclosure in Malaysia Following the Economic Crisis. *Journal of International Accounting, Auditing and Taxation*, 15(2), 226-248
- Mossin, J. (1966). Equilibrium in a Capital Asset Market. *Econometrica*, 34(4), 768-783.
- Mouritsen, J., Bukh, P., Flagstad, K., Thorbjørnsen, S., Johansen, M., Kotis, S., et al. (2003). Intellectual Capital Statements-the New Guideline. Denmark: Danish Ministry of Science, Technology and Innovation Cph.
- Muijs, D. (2004). *Doing Quantitative Research in Education with SPSS*. London: SAGE Publications Lnc.
- Nachmias, C. F. and Nachmias, D. (1996). *Reseach Methods in Social Scince*. London: Edward Arnold.
- Nash, H. (1998). Accounting for the Future, a Discplined Approach to Value-Added Accounting. Available at:

 http://home.sprintmail.com/~humphreynash/The_AIMR_and_AFTF.htm
 (accessed 30 September 2012).
- Nasseh, A. and Strauss, J. (2004). Stock Prices and the Dividend Discount Model: Did Their Relation Break Down in the 1990s? *The Quarterly Review of Economics and Finance*, 44(2), 191-207.

- Neter, J., Kutner, M. H. and Nachtsheim, C. (2003). *Applied Linear Regression Models*. Chicago: Irwin.
- Neter, J., Wasserman, W. and Kutner, M. (1983). *Applied Linear Regression Models*. Homewood, IL: Richard D. Irwin.
- Ng, J. (2008). The Effect of Information Quality on Liquidity Risk. *Journal of Accounting and Economics*, 52(2/3), 126-143.
- Nielsen, C., Bukh, P. N., Mouritsen, J., Johansen, M. R. and Gormsen, P. (2006). Intellectual Capital Statements on Their Way to The Stock Exchange Analyzing New Reporting Systems. *Journal of Intellectual Capital*, 7(2), 1469-1930.
- OECD. (1999). see Organisation for Economic Co-operation and Development.
- OECD. (2006). Creating Value From Intellectual Assets. Paris: Paper Presented at Meeting of the OECD Council at Ministerial Level.
- Ogier, T., Rugman, J. and Spicer, L. (2004). *The Real Cost of Capital* (1st ed.). Great Britain: FT Prentice Hall.
- Ohlson, J. A. (1995). Earnings, Book Values, and Dividends in Equity Valuation. Contemporary Accounting Research, 11(2), 661-687.
- Ohlson, J. A. and Juettner-Nauroth, B. E. (2005). Expected EPS and EPS Growth as Determinants of Value. *Review of Accounting Studies*, 10(2/3), 349-365.
- Ohlson, J. A. and Zhang, X.-J. (1998). Accrual Accounting and Equity Valuation. *Accounting Research*, 36(3), 85-111.
- Orens, R., Aerts, W. and Lybaert, N. (2009). Intellectual Capital Disclosure, Cost of Finance and Firm Value. *Management Decision*, 47(10), 1536-1554.
- Organisition for Economic Co-operation and Development (OECD). (1999).

 "Guidelines and Instruction for OECD Symposium". International Symposium Measuring and Reporting Intellectual Capital: Experience, Issues, and Prospects. Amsterdam. Paris: OECD.
- Ota, K. (2001). The Value-Relevance of Book Value, Current Earnings, and Management Forecasts of Earnings. *Working Paper. Australian National University*.

- Pae, J., Thornton, D. B. and Welker, M. (2005). The Link between Earnings Conservatism and The Price-to-Book Ratio. *Contemporary Accounting Research*, 22(3), 693-717.
- Passetti, E., Tenucci, A., Cinquini, L. and Frey, M. (2008). Communicating Intellectual Capital: Evidence from Social and Sustainability Reporting. *Working Paper*, University Library of Munich, Germany.
- Penman, S. H. (1992). Return to Fundamentals. *Journal of Accounting, Auditing and Finance*, 7(4), 465-484.
- Penman, S. H. (2003). The Quality of Financial Statements: Perspectives from the Recent Stock Market Bubble. *Accounting Horizons*, 17(Supplement), 77-96.
- Penman, S. H. and Zhang, X.-J. (2002). Accounting Conservatism, the Quality of Earnings, and Stock Returns. *The Accounting Review*, 77(2), 237-264.
- Petruska, K. A. (2008). Accounting Conservatism, Cost of Capital, and Fraudulent Financial Reporting. PhD Thesis, Kent State University, Kent.
- Petty, R. and Guthrie, J. (2000). Intellectual Capital Literature Review: Measurement, Reporting and Management. *Journal of Intellectual Capital*, 1(2), 155-176.
- Phillips, J. J. and Phillips, P. P. (2002). *Measuring Intellectual Capital*. United States of America: ASTD.
- Pike, S., Rylander, A. and Roos, G. (2002). "Intellectual Capital Management and Disclosure", in Choo, C.W. and Bontis, N. (Eds), The Strategic Management of Intellectual Capital and Organizational Knowledge, Oxford University Press, New York, NY.
- Planting, S. (2000). A Fever for Talent. Available at:http://secure.financialmail.co.za/00/0505/covopin/cover.htm (accessed 30 September 2012).
- Pulic, A. (1998). Measuring the Performance of Intellectual Potential in Knowledge Economy. *Available at: www.vaic-on.net (accessed 30 September 2012)*.
- Pulic, A. (2004). Intellectual Capital-Does It Create or Destroy Value? *Measuring Business Excellence*, 8(1), 62-68.

- Rahman, A. F. and Mohd-Saleh, N. (2008). The Effect of Free Crash Flow Agency Problem on the Value Relevance of Earnings and Book Value. *Financial Reporting & Accounting*, 6(1), 75-90.
- Ramesh, K. and Thiagarajan, R. (1995). Inter-Temporal Decline in Earnings Response Coefficients. Working Paper, Northwestern University, Evanston, IL.
- Rapp, M. S. (2010). Information Asymmetries and the Value-Relevance of Cash Flow and Accounting Figures-Empirical Analysis and Implications for Managerial Accounting. *Problems and Perspectives in Management*, 8(2), 64-75.
- Reed, S., Capell, K., Reinhardt, A. and Matlack, C. (2002). Searching for Enrons. *Business Week*, 25 February, 24-25.
- Reinita, I. S. M. (2007). *Value relevance of R&D reporting*. Master Thesis. University of Maastricht Faculty of Economics and Business Administration.
- Richardson, A. J. and Welker, M. (2001). Social Disclosure, Financial Disclosure and the Cost of Equity Capital. *Accounting, Organizations and Society*, 26(7), 597-616.
- Richardson, S. A., Sloan, R. G., Soliman, M. T. and Tuna, I. (2005). Accrual Reliability, Earnings Persistence and Stock Prices. *Journal of Accounting & Economics*, 39(3), 437-485.
- Robb, S. W. G., Single, L. E. and Zarzeski, M. T. (2001). Nonfinancial Disclosures Across Anglo-American Countries. *Journal of International Accounting*, *Auditing & Taxation*, 10(1), 71-83.
- Roos, G., Pike, S. and Fernstrom, L. (2005). *Managing Intellectual Capital in Practice*. New York: Butterworth-Heinemann.
- Roos, J., Roos, G., Dragonetti, N. C. and Edvinsson, L. (1997). *Intellectual Capital:* Navigating in the New Business Landscape. London: Macmillan.
- Ross, S. A. (1976). The Arbitrage Theory of Capital Asset Pricing. *Journal of Economic Theory*, 13(3), 341-360.

- Ross, S. A. (1977). The Determination of Financial Structure: the Incentive-Signalling Approach. *The Bell Journal of Economics*, 8(1), 23-40.
- Ross, S. A., Westerfield, R. W. and Jordan, B. D. (2010). *Fundamentals of Corporate Finance* (9th ed.). New York: McGraw-Hill/Irwin.
- Rucker, D. D., Preacher, K. J., Tormala, Z. L. and Petty, R. E. (2011). Mediation Analysis in Social Psychology: Current Practices and New Recommendations. Social and Personality Psychology Compass, 5(6), 359–371.
- Rylander, A., Jacobsen, K. and Roos, G. (2000). Towards Improved Information Disclosure on Intellectual Capital. *International Journal of Technology Management*, 20(5), 715-741.
- Saint-Onge, H. (1996). Tacit Knowledge: The Key to the Strategic Alignment of Intellectual Capital. *Strategy and Leadership*, 24(2), 10-16.
- Schipper, K. and Vincent, L. (2003). Earnings Quality. *Accounting Horizons*, 17(Supplement), 97-110.
- Scott, W. R. (2003). Financial Accounting Theory (3rd ed.). Toronto: Prentice Hall.
- Shamki, D. and Rahman, A. A. (2012). Value Relevance of Earnings and Book Value: Evidence from Jordan. *International Journal of Business and Management*, 7(10), 133-141.
- Sharpe, W. F. (1964). Capital Asset Prices: A Theory of Market Equilibrium under Conditions of Risk. *The Journal of Finance*, 19(3), 425-442.
- Shi, C. (2002). Accounting for Intangibles, Earnings Variability and Analysts' Forecasts: Evidence from the Software Industry. *Working Paper. Graduate School of Management, University of California*.
- Shiu, H.-J. (2006). The Application of the Value Added Intellectual Coefficient to Measure Corporate Performance: Evidence from Technological Firms. *International Journal of Management*, 23(2), 356-365.
- Singh, I. and Zahn, J.-L. W. M. V. d. (2007). Does Intellectual Capital Disclosure Reduce an IPO's Cost of Capital? *Journal of Intellectual Capital*, 8(3), 494-516.

- Singh, S. and Kansal, M. (2011). Voluntary Disclosures of Intellectual Capital: An Empirical Analysis. *Journal of Intellectual Capital*, 12(2), 301-318.
- Skandia. (1994). Visualising Intellectual Capital in Skandia. A Supplement to Skandia's 1994 Annual Report. Sweden: Skandia.
- SKE. (2005). see Society for Knowledge Economics.
- Society for Knowledge Economics (SKE). (2005). Australian Guiding Principles on Extended Performance Management: A Guide to Better Managing, Measuring and Reporting Knowledge Intensive Organizational Resources.

 GAP Congress on Knowledge Capital, Parliament House, Melbourne: SKE.
- Sofian, S. (2005). *Intellectual Capital and Management Acoounting Practices*. PhD Thesis, University of Bradford, Bradford.
- Sonnier, B. M. (2008). Intellectual Capital Disclosure: High-Tech Versus Traditional Sector Companies. *Journal of Intellectual Capital*, 9(4), 705-722.
- Sorensen, E. H. and Williamson, D. A. (1985). Some Evidence on the Value of Dividend Discount Models. *Financial Analysts Journal*, 41(6), 60-69.
- Spence, M. (1973). Job Market Signaling. *The Quarterly Journal of Economic*, 87(3), 355-374.
- Spence, M. (2002). Signaling in Retrospect and the Informational Structure of Markets. *American Economic Review* 92(3), 434-459.
- Standfield, K. (1998). Extending the Intellectual Capital Framework. *Available at:* www.knowcorp.com/article075.com (accessed 30 September 2012).
- Stedinger, J. R. and Tasker, G. D. (1985). Regional Hydrologic Analysis: 1.
 Ordinary, Weighted, and Generalized Least Squares Compared. Water
 Resources Research, 21(9), 1421–1432.
- Stedinger, J. R. and Tasker, G. D. (1986a). Correction to Regional Hydrologic Analysis, 1.Ordinary, Weighted, and Generalised Least Squares Compared. Water Resources Research, 22(5), 844.
- Stedinger, J. R. and Tasker, G. D. (1986b). Regional Hydrologic Analysis, 2. Model Error Estimators, Estimation of Sigma and Log-Pearson Type 3 Distributions. *Water Resources Research*, 22(10), 1487-1499.

- Stewart, T. A. (1997). *Intellectual capital: The New Wealth of Organizations*. New York: Bantam Doubleday Dell Publishing Group.
- Stickney, C. P., Weil, R. L., Schipper, K. and Francis, J. (2009). *Financial Accounting: an Introduction to Concepts, Methods and Uses* (13th ed.). U.S.A: Soth-Western Cengage Learning.
- Stober, T. L. (1996). Do Prices Behave as if Accounting Book Values Are Conservative? Cross-Sectional Tests of the Feltham -Ohlson [1995] Valuation Model. Working Paper retrieved from
- Stubelj, I. (2010). The Cost of Equity Capital on Developing Equity Markets: Estimations for Selected Slovene Companies. *Journal of Economics & Business*, 5(1), 67-74.
- Suazo, M. M., Martínez, P. G. and Sandoval, R. (2009). Creating Psychological and Legal Contracts Through Human Resource Practices: A Signaling Theory Perspective. *Human Resource Management Review*, 19(2), 154-166.
- Sullivan, P. H. (1998). Profiting from Intellectual Capital: Extracting Value from Innovation. New York: John Wiley & Sons Inc.
- Sullivan, P. H. (1999). Profiting from Intellectual Capital. *Journal of Knowledge Management*, 3(2), 132-143.
- Sullivan, P. H. (2000). Value-Driven Intellectual Capital. How to Convert Intangible Corporate Assets into Market Value. New York: J. Wiley & Sons.
- Sveiby, K. E. (1997). The New Organizational Wealth: Managing & Measuring Knowledge-Based Assets. San Francisco: Berrett-Koehler Pub.
- Tham, J. and Velez-Pareja, I. (2004). Principles of Cash Flow Valuation: An Integrated Market Based Approach. USA: Elsevier Academic Press.
- Thinggaard, F. and Damkier, J. (2008). Has Financial Statement Information Become Less Relevant? Longitudinal Evidence from Denmark. *Scandinavian Journal of Management*, 24(4), 375-387.
- Ting, I. W. K. and Lean, H. H. (2009). Intellectual Capital Performance of Financial Institutions in Malaysia. *Journal of Intellectual Capital*, 10(4), 588-599.

- Tokmakcioglu, K., Tas, O. and Gozlu, S. (2007). Intangible Assets and Cost of Capital: An Application for a Technology Utilizing Firm. *Paper presented at the PICMET*, Portland, Oregon USA.
- Trochim, W. M. K. (2000). *The Research Methods Knowledge Base* (2nd ed.). Cincinnati, OH: Atomic Dog Publishing.
- Tukey, J. W. (1977). Exploratory Data Analysis. New York: Addison-Wesley.
- Udayashantha, P. D. C. (2006). Intangible Assets: An Assessment of Application of Generally Accepted Accounting Principles and Accounting Standards for Recognition and Measurements of Intangible Assets. *Paper presented at the Business And Information*, Singapore.
- Vafaei, A., Taylor, D. and Ahmed, K. (2011). The Value Relevance of Intellectual Capital Disclosures. *Journal of Intellectual Capital*, 12(3), 407-429.
- Vazquez, R. D., Valdes, A. L. and Herrera, H. V. (2007). Value Relevance of the Ohlson Model with Mexican Data. *Working Paper. Universidad Nacional Autonma de Mexico*.
- Verdi, R. S. (2008). Information Environment and the Cost of Equity Capital. Working Paper. University of Pennsylvania.
- Wall, T. (2002). Mental Arithmetic. *CIMA Financial Management*, December/January, 28-29.
- Wallman, S. M. H. (1995). The Future of Accounting and Disclosure in an Evolving World: the Need for Dramatic Change. *Accounting Horizons*, 9(3), 81-91.
- Wang, M. C. (2013). Value Relevance on Intellectual Capital Valuation Methods: the Role of Corporate Governance. *Quality and Quantity*, 47(2), 1213–1223.
- Warfield, T. D. and Wild, J. J. (1992). Accounting Recognition and the Relevance of Earnings as an Explanatory Variable for Returns. *The Accounting Review*, 67(4), 821-842.
- Watts, R. L. and Zimmerman, J. L. (1978). Towards a Positive Theory of the Determination of Accounting Standards. *The Accounting Review*, 53(1), 112-134.

- Watts, R. L. and Zimmerman, J. L. (1986). *Positive Accounting Theory*. London: Prentice-Hall.
- Welker, M. (1995). Disclosure Policy, Information Asymmetry, and Liquidity in Equity Markets. *Contemporary Accounting Research*, 11(2), 801-827.
- Williams, S. M. (2001). Is Intellectual Capital Performance and Disclosure Practices Related? *Journal of Intellectual Capital*, 2(3), 192-203.
- Wong, L. (2008). *Earnings Quality and the Cost of Equity in Australia*. PhD Thesis, University of New South Wales, Sydney.
- Wooldridge, J. M. (2002). *Econometric Analysis of Cross Section and Panel Data*. Cambridge, MA: MIT Press.
- World Bank. (2005). World Development Indicator. *Available at:* http://www.worldbank.org/data/wdi2005/wditext/section.htm.
- Wright, P. M., McMahan, G. C., McCormick, B. and Sherman, W. S. (1998).
 Strategy, Core Competence, and HR Involvement as Determinants of HR
 Effectiveness and Refinery Performance. *Human Resource Management*,
 37(1), 17-29.
- Wyatt, A. and Frick, H. (2010). Accounting for Investments in Human Capital: A Review. *Australian Accounting Review*, 54(20), 199-220.
- Yalama, A. and Coskun, M. (2007). Intellectual Capital Performance of Quoted Banks on the Istanbul Stock Exchange Market. *Journal of Intellectual Capital*, 8(2), 256-271.
- Youndt, M. A. (1998). *Human Resource Management System, Intellectual Capital and Organizational Performance*. PhD Thesis, Pennsylvania State University, Pennsylvania.
- Yu, H.-C., Wang, W.-Y. and Chang, C. (2009). The Pricing of Intellectual Capital in the It Industry. *Available at SSRN: http://ssrn.com/abstract=1327668 or http://dx.doi.org/10.2139/ssrn.1327668 (accessed 30 September 2012)*.
- Zaimee, A. (2007). Mini-ENRONS Shaking Up Malaysia's Corporate Governance? *Accountants Today*, 20(July), 20-23.

- Zambon, S. (2003). Study on the Measurement of Intangible Assets and Associated Reporting Practices. Brussels: Commission of the European Communities Enterprise Directorate General.
- Ze'ghal, D. and Maaloul, A. (2010). Analysing Value Added as an Indicator of Intellectual Capital and Its Consequences on Company Performance. *Journal of Intellectual Capital*, 11(1), 39-60.
- Zhang, Y. and Wiersema, M. F. (2009). Stock Market Reaction to CEO Certification: The Signaling Role of CEO Background. *Strategic Management Journal*, 30(7), 693-710.
- Zohar, D. and Marshall, I. (2001). *SQ: Connecting with Our Spiritual Intelligence*. New York: Bloomsbury Publishing.
- Zohar, D. and Marshall, I. (2004). *Spiritual Capital: Wealth We Can Live By*. U.S.A: Bloomsbury Publishing Plc.

TABLE OF CONTENTS

CHAPTER		TITLE	PAGE
	DECI	LARATION	ii
	DEDI	CATION	iii
	ACK	NOWLEDGMENT	iv
	ABST	TRACT	V
	ABST	TRAK	vi
	TABI	LE OF CONTENTS	vii
	LIST	OF TABLES	xvi
	LIST	OF FIGURES	xxiii
	LIST	OF ABBREVIATIONS	XXV
	LIST	OF APPENDICES	xxviii
1	INTR	ODUCTION	1
	1.1	Overview	1
	1.2	Statement of the Problem	5
	1.3	Purpose of the Study	10
	1.4	Objectives of the Study	10
	1.5	Significance of the Study	11
	1.6	Scope of the Study	12
	1.7	Outline of the Thesis	12
	1.8	Terminologies	13

	1.9	Summa	ary	15
2	EAR	NINGS A	ATTRIBUTES AND COST OF EQUITY	16
	2.1	Introdu	action	16
	2.2	Earning	gs Quality	18
		2.2.1	Constructs of Earnings Quality	19
			2.2.1.1 Constructs of Earnings Quality- Associations between Income, Cash Flows and Accruals	20
			2.2.1.2 Earnings Quality Structures –Time Series Attributes of Earnings	23
			2.2.1.3 Earnings Quality Structures – FASB No. 2	25
			2.2.1.4 Earnings Quality Structures- Implementation Decisions	26
		2.2.2	Relevance and Conservatism of Earnings	27
			2.2.2.1 Value Relevance of Earnings	28
			2.2.2.2 Conservatism of Earnings	33
	2.3	Introdu	action to Cost of Capital	37
		2.3.1	Significance of Cost of Capital	38
		2.3.2	Capital	38
		2.3.3	Cost of Capital	39
		2.3.4	Costs of Capital Structure	40
			2.3.4.1 Cost of Common Equity	41
		2.3.5	Prior Literature on Implied CoE	44
	2.4	Prior L	iterature on Attributes of Earnings and CoE	50
	2.5		ary of Past Researches on Relationship of	53

	2.6	Summa	ary	56			
3	ACC	INTELLECTAL CAPITAL, RELEVANCE OF ACCOUNTING INFORMATION AND COST OF EQUITY					
	3.1	Introdu	action	58			
	3.2	Intelled	ctual Capital	60			
		3.2.1	IC and Capital Market	62			
		3.2.2	Definition and Classification of IC	63			
			3.2.2.1 Human Capital	66			
			3.2.2.2 Structural Capital	67			
			3.2.2.3 Relational Capital	67			
			3.2.2.4 Spiritual Capital	68			
		3.2.3	Models/Frameworks of IC	69			
			3.2.3.1 Skandia Model	69			
			3.2.3.2 Brooking's Model	70			
			3.2.3.3 Sveiby Model	71			
			3.2.3.4 Bontis Model	73			
			3.2.3.5 Pulic (VAIC) Model	74			
			3.2.3.6 Ismail Model	75			
		3.2.4	Traditional Financial Reporting and Its Limitations	76			
		3.2.5	IC Reporting	77			
		3.2.6	Measurement IC	78			
	3.3	IC and	Relevance of Accounting Information	87			
	3.4	Conser	vatism of IC and Accounting Information	92			
	3.5	IC and	CoE	99			

	3.6	Summa	ary of Related Researches	103
	3.7	Summa	ary	108
4	THE	ORETIC	CAL FRAMEWORK	109
	4.1	Introdu	action	109
	4.2	Theori	es Related to the Study	109
		4.2.1	Signalling Theory	110
		4.2.2	Valuation Theory	112
	4.3	Theore	etical Frameworks and Hypotheses	113
		4.3.1	Framework I (Accounting Earnings Attributes and CoE)	114
			4.3.1.1 Independent Variables	115
			4.3.1.2 Dependent Variable	117
			4.3.1.3 Control Variables	118
			4.3.1.4 Hypotheses Development (Framework I - Accounting Earnings Attributes and CoE)	120
		4.3.2	Framework II: Effect of Intellectual Capital (IC) as the Moderator on the Relationship between Accounting Earnings and Book Value (Individually and Aggregate) on	
			Market Price	121
			4.3.2.1 Independent Variables	122
			4.3.2.2 Moderating Variables	122
			4.3.2.3 Dependent Variable	124
			4.3.2.4 Hypotheses Development (Framework II - Effect of Intellectual Capital (IC) as the Moderator on the Relationship	

			between Accounting Earnings and	
			Book Value (Individually and	
			Aggregate) on Market Price)	126
		4.3.3	Framework III (Effect of IC on	
			Relationship between Relevance of	
			Earnings and CoE)	128
			4.3.3.1 Hypotheses Development	
			(Framework III - Effect of IC on	
			Relationship between Relevance of	
			Earnings and CoE)	129
	4.4	Summ	ary	130
_		~		
5	REA	SERCH	METHODOLOGY	133
	5.1	Introdu	uction	133
	5.2	Resear	rch Process	133
	5.3	Metho	dology	134
		5.3.1	Sampling Technique	137
		5.3.2	Data Collection	140
		5.3.3	Data Analysis	140
			5.3.3.1 Descriptive Statistics	140
			5.3.3.2 Correlation Analysis	141
			5.3.3.3 Multiple Regression and	
			Moderated Multiple Regression	
			Analysis	142
			5.3.3.4 Further Analysis and Robustness	
			Test	148
		5.3.4	Assumptions for Multiple Regression	148
	5.4	Summ	ary	150

6	FIND	INGS			154
	6.1	Introduc	ction		154
	6.2	Applyin	g the A	ssumptions of Linear Model of	
		Multiple	e Regressi	on Analysis	155
	6.3	Descript	tive Statis	tics	157
	6.4	Correlat	ion Matri	K	161
	6.5	_	ion Analy of Equity	sis I: Effect of Earnings Attributes	169
		6.5.1	Effect of Equity	Relevance of Earnings on Cost of	169
		6.5.2	Effect of of Equity	Conservatism of Earnings on Cost	172
		6.5.3	Addition	al Analyses and Robustness Tests	174
			6.5.3.1	Different Regression Estimators	175
			6.5.3.2	Additional Control Variables	178
	6.6	Regress	ion Analy	sis II: Effect of Intellectual Capital	
		` /		rator on the Relationship between	
			C	ngs and Book Value (Individually	100
			_	n Market Price	180
		6.6.1	Moderato	or on the Relationship between	
				d Accounting Earnings and Book Market Price	180
				Effect of Human Capital (HC) as the Moderator on the Relationship between Combined Accounting Earnings and Book Value on Market Price	181
				Effect of Structural Capital (SC) as the Moderator on the Relationship	

		Earnings and Book Value on	
		Market Price	185
	6.6.1.3	Effect of Intellectual Capital (IC) as the Moderator on the Relationship between Combined Accounting Earnings and Book	
		Value on Market Price	189
	6.6.1.4	Additional Analyses and Robustness Tests through	
		Different Regression Estimators	193
6.6.2		of Intellectual Capital (IC) as the tor on the Relationship between	
	Accoun	ting Earnings and Market Price	198
	6.6.2.1	Effect of Human Capital (HC) as the Moderator on the Relationship between Accounting Earnings and	
		Market Price	198
	6.6.2.2	Effect of Structural Capital (SC) as the Moderator on the Relationship between Accounting Earnings and	
		Market Price	202
	6.6.2.3	Effect of Intellectual Capital (IC) as the Moderator on the Relation between Accounting Earnings and	
		Market Price	205
	6.6.2.4	Additional Analyses and Robustness Tests through	
		Different Regression Estimators	209

	6.6.3	Effect of Intellectual Capital (IC) as the	
		Moderator on the Relationship between Book Value and Market Price	213
		6.6.3.1 Effect of Human Capital (HC) as the Moderator on the Relationship between Book Value and Market Price	213
		6.6.3.2 Effect of Structural Capital (SC) as the Moderator on the Relationship between Book Value and Market Price	217
		6.6.3.3 Effect of Intellectual Capital (IC) as the Moderator on the Relationship between Book Value	
		and Market Price 6.6.3.4 Additional Analyses and Robustness Tests through	221 225
6.7	_	Different Regression Estimators sion Analysis III: Effect of Intellectual I (IC) as the Moderator on the Relationship	223
	betwee 6.7.1	en Relevance of Earnings and Cost of Equity Effect of Human Capital (HC) as the Moderator on the Relationship between Relevance of Earnings and Cost of Equity	229 229
	6.7.2	Effect of Structural Capital (SC) as the Moderator on the Relationship between Relevance of Earnings and Cost of Equity	235
	6.7.3	Effect of Intellectual Capital (IC) as the Moderator on the Relationship between Relevance of Earnings and Cost of Equity	239
	6.7.4	Additional Analyses and Robustness Tests	243

			6.7.4.1 I	Different Regressio	n Estimators	244
			6.7.4.2 A	Additional Control	Variables	246
	6.8	Summa	ry			250
7	DISC	USSION	AND CO	NCLUSION		254
	7.1	Introduc	ction			254
	7.2	Overvie	w of the St	tudy		254
	7.3	Discuss	ion of Find	ings		255
		7.3.1	Association and Cost	on between Earn	ings Attributes	256
		7.3.2	Moderato Accounting	f Intellectual C r on the Relationg ng Earnings and ally and Aggrega	nship between Book Value	259
		7.3.3	Moderato	f Intellectual Car on the Relation of Earnings and C	nship between	262
	7.4	Conclus	ion			264
	7.5	Theory	of Study			265
		7.5.1	Valuation	Theory		266
		7.5.2	Signalling	g Theory		268
	7.6	Limitati	on of Stud	y		270
	7.7	Contrib	utions of S	tudy		271
	7.8	Implicat	tions of the	Study		273
	7.9	Recomm	nendations	for Future Research	ch	275
REFERENC	CES					277
Appendices .	A-B				3	307-314

LIST OF TABLES

TABLE NO.	TITLE						
2.1	Summary of Relevant Research	54					
3.1	Economic Transformations	61					
3.2	Summary of Definition and Dimension of IC	64					
3.3	Intangible Assets Monitor of Celemi	72					
3.4	Advantages and Disadvantages of IC Measuring						
	Methods	80					
3.5	Summarizes the Classification of the IC Measurement						
	Models	81					
3.6	Findings of Collins et al. (1997) Study	88					
3.7	Summary of Past Researches	103					
4.1	The Process of Calculating VAIC	123					
4.2	Summary of Hypotheses Development Related to						
	Framework I	130					
4.3	Summary of Hypotheses Development Related to						
	Framework II	131					
4.4	Summary of Hypotheses Development Related to						
	Framework III	132					
5.1	Number of Companies According to Industries on Main						
	Board of Bursa Malaysia for 2010	138					
5.2	Company-Observations of Framework I	139					
5.3	Company-Observations of Framework II and III	139					
5.4	Rule of Thumb on Correlation Coefficient Size	141					
5.5	Summary of all Variables	151					
6.1	Normal Data Distribution before and after						
	Transformation Related to Framework I	155					

6.2	Normal Data Distribution before and after						
	Transformation Related to Framework II	156					
6.3	Normal Data Distribution before and after						
	Transformation Related to Framework III	156					
6.4	Normal Data Distribution before and after						
	Transformation Related to Framework III	157					
6.5	Descriptive Statistics Analysis Result Related to						
	Framework I	158					
6.6	Descriptive Statistics Analysis Result Related to						
	Framework II	159					
6.7	Descriptive Statistics Analysis Result Related to						
	Framework III	160					
6.8	Pairwise Correlation Matrix Related to Framework I	161					
6.9	Pairwise Correlation Matrix Related to Framework II						
	(HCE as Moderator)	162					
6.10	Pairwise Correlation Matrix Related to Framework II						
	(SCE as Moderator)	163					
6.11	Pairwise Correlation Matrix Related to Framework II						
	(ICE as Moderator)	164					
6.12	Pairwise Correlation Matrix Related to Framework III						
	(HC as Moderator)	165					
6.13	Pairwise Correlation Matrix Related to Framework III						
	(SC as Moderator)	166					
6.14	Pairwise Correlation Matrix Related to Framework III						
	(IC as Moderator)	167					
6.15	VIF and Tolerance Tests	170					
6.16	Heteroscedasticity Test	171					
6.17	The Result of OLS Regression to Examine the						
	Association between Relevance of Earnings and Cost of						
	Equity	171					
6.18	VIF and Tolerance Tests	173					
6.19	Heteroscedasticity Test	173					

6.20	The Result of OLS Regression to Examine the	
	Association between Conservatism of Earnings and Cost	
	of Equity	174
6.21	The Results of Different Regressions Estimators Related	
	to Examine the Association between Relevance of	
	Earnings and Cost of Equity	176
6.22	The Results of Different Regressions Estimators Related	
	to Examine the Association between Conservatism of	
	Earnings and Cost of Equity	177
6.23	The Result of OLS Regression to Examine the	
	Association between Relevance of Earnings and Cost of	
	Equity with the Additional Control Variable	178
6.24	The Result of OLS Regression to Examine the	
	Association between Conservatism of Earnings and Cost	
	of Equity with the Additional Control Variable	179
6.25	VIF and Tolerance Value Related to HC as Moderator	183
6.26	Heteroscedasticity Test Related to HC as Moderator	183
6.27	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of HC on Relationship between	
	Earnings, Book Value and Market Price	184
6.28	VIF and Tolerance Value Related to SC as Moderator	187
6.29	Heteroscedasticity Test Related to SC as Moderator	187
6.30	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of SC on Relationship between	
	Earnings, Book Value and Market Price	188
6.31	VIF and Tolerance Value Related to IC as Moderator	191
6.32	Heteroscedasticity Test Related to IC as Moderator	191
6.33	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of IC on Relationship between	
	Earnings, Book Value and Market Price	192
6.34	The Result of Different Regression Estimators Related to	
	Effect of HC on Relationship between Earnings, Book	
	Value and Market Price	195

6.35	The Result of Different Regression Estimators Related to	
	Effect of SC on Relationship between Earnings, Book	
	Value and Market Price	196
6.36	The Result of Different Regression Estimators Related to	
	Effect of IC on Relationship between Earnings, Book	
	Value and Market Price	197
6.37	VIF and Tolerance Value Related to HC as Moderator	200
6.38	Heteroscedasticity Test Related to HC as Moderator	200
6.39	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of HC on Relationship between	
	Earnings and Market Price	201
6.40	VIF and Tolerance Value Related to SC as Moderator	203
6.41	Heteroscedasticity Test Related to SC as Moderator	204
6.42	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of SC on Relationship between	
	Earnings and Market Price	204
6.43	VIF and Tolerance Value Related to IC as Moderator	207
6.44	Heteroscedasticity Test Related to IC as Moderator	207
6.45	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of IC on Relationship between	
	Earnings and Market Price	208
6.46	The Result of Different Regression Estimators Related to	
	Effect of HC on Relationship between Earnings and	
	Market Price	210
6.47	The Result of Different Regression Estimators Related to	
	Effect of SC on Relationship between Earnings and	
	Market Price	211
6.48	The Result of Different Regression Estimators Related to	
	Effect of IC on Relationship between Earnings and	
	Market Price	212
6.49	VIF and Tolerance Value Related to HC as Moderator	215
6.50	Heteroscedasticity Test Related to HC as Moderator	215

6.51	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of HC on Relationship between Book	
	Value and Market Price	216
6.52	VIF and Tolerance Value Related to SC as Moderator	219
6.53	Heteroscedasticity Test Related to SC as Moderator	219
6.54	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of SC on Relationship between Book	
	Value and Market Price	220
6.55	VIF and Tolerance Value Related to IC as Moderator	223
6.56	Heteroscedasticity Test Related to IC as Moderator	223
6.57	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of IC on Relationship between Book	
	Value and Market Price	224
6.58	The Result of Different Regression Estimators Related to	
	Effect of HC on Relationship between Book Value and	
	Market Price	226
6.59	The Result of Different Regression Estimators Related to	
	Effect of SC on Relationship between Book Value and	
	Market Price	227
6.60	The Result of Different Regression Estimators Related to	
	Effect of IC on Relationship between Book Value and	
	Market Price	228
6.61	VIF and Tolerance Value Related to Framework III	
	(Moderating of HC)	232
6.62	Heteroscedasticity Test Related to Framework III	
	(Moderating of HC)	232
6.63	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of HC on Relationship between	
	Relevance of Earnings and Cost of Equity	233
6.64	VIF and Tolerance Value Related to Framework III	
	(Moderating of SC)	236
6.65	Heteroscedasticity Test Related to Framework III	
	(Moderating of SC)	237

6.66	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of SC on Relationship between	
	Relevance of Earnings and Cost of Equity	238
6.67	VIF and Tolerance Value Related to Framework III	
	(Moderating of IC)	240
6.68	Heteroscedasticity Test Related to Framework III	
	(Moderating of IC)	241
6.69	The Result of Hierarchical Multiple Regression to	
	Examine the Effect of IC on Relationship between	
	Relevance of Earnings and Cost of Equity	242
6.70	The Result of Different Regression Estimator Related to	
	Examine the Effect of HC on the Relationship between	
	Relevance of Earnings and Cost of Equity	244
6.71	The Result of Different Regression Estimator Related to	
	Examine the Effect of SC on the Relationship between	
	Relevance of Earnings and Cost of Equity	245
6.72	The Result of Different Regression Estimator Related to	
	Examine the Effect of IC on the Relationship between	
	Relevance of Earnings and Cost of Equity	246
6.73	The Result of OLS Regression to Examine the Effect of	
	HC on the Relationship between Relevance of Earnings	
	and Cost of Equity with Additional Control Variable	247
6.74	The Result of OLS Regression to Examine the Effect of	
	SC on the Relationship between Relevance of Earnings	
	and Cost of Equity with Additional Control Variable	248
6.75	The Result of OLS Regression to Examine the Effect of	
	IC on the Relationship between Relevance of Earnings	
	and Cost of Equity with Additional Control Variable	249
6.76	Summary of Hypotheses Testing Findings Related to	
	Framework I	251
6.77	Summary of Hypotheses Testing Findings Related to	
	Framework II	252
6.78	Summary of Hypotheses Testing Findings Related to	
	Framework III	253

7.1 Legatum Prosperity Index

258

LIST OF FIGURES

FIGURE NO	. TITLE	PAGE
2.1	Types of Income Smoothness	24
3.1	Economic Eras	60
3.2	Skandia Value Model	70
3.3	Brooking's IC Model	71
3.4	Bontis' Intangible Capital Model	73
3.5	Overview of the VAIC Model	74
3.6	The "Intangible Capital Central-Triangle" Framework	75
3.7	Motivation to Measure and Report IC	77
3.8	GAAP Treatment for Intangible Assets	95
3.9	Different Accounting Treatments and Decision	
	Usefulness	95
4.1	Signaling Timeline	110
4.2	Research Framework I for Earnings Attributes and CoE	119
4.3	Framework II (the Effect of IC and its Components (HC	
	and SC) on the Relationship between Accounting	
	Earnings, Book Value and Market Value)	124
4.4	Framework II (the Effect of IC and its Components (HC	
	and SC) on the Relationship between Accounting	
	Earnings and Market Value)	125
4.5	Framework II (the Effect of IC and its Components (HC	
	and SC) on the Relationship between Book Value and	
	Market Value	125

4.6	Framework III (the Effect of IC and its Components (HC	
	and SC) on Relationship between Relevance of Earnings	
	and CoE)	128
5.1	Research Process	135
5.2	Knowledge Claims, Strategies of Inquiry, and Methods	
	Leading to Approaches and the Design Process	136
5.3	Moderator Model	144
6.1	The Moderating Effect of Human Capital on EPS, BVPS	
	and Stock Price	185
6.2	The Moderating Effect of Structural Capital on EPS,	
	BVPS and Stock Price	189
6.3	The Moderating Effect of Intellectual Capital on EPS,	
	BVPS and Stock Price	193
6.4	The Moderating Effect of Human Capital on EPS and	
	Stock Price	202
6.5	The Moderating Effect of Structural Capital on EPS and	
	Stock Price	205
6.6	The Moderating Effect of Intellectual Capital on EPS	
	and Stock Price	209
6.7	The Moderating Effect of Human Capital on BVPS and	
	Stock Price	217
6.8	The Moderating Effect of Structural Capital on BVPS	
	and Stock Price	221
6.9	The Moderating Effect of Intellectual Capital on BVPS	
	and Stock Price	225
6.10	The Moderating Effect of Human Capital on Relevance	
	of Earnings and CoE	234
6.11	The Moderating Effect of Intellectual Capital on	
	Relevance of Earnings and CoE	243

LIST OF ABBREVIATIONS

AICPA - American Institute of Certified Public Accountants

BV - Book Value of Equity

BVPSHCE - Interaction of Book Value Per Share and Human Capital

Efficiency

BVPSSCE - Interaction of Book Value Per Share and Structural Capital

Efficiency

BVPSICE - Interaction of Book Value Per Share and Intellectual Capital

Efficiency

B/M - Book Value of Equity to Market Value of Equity

BSC - Balanced Scorecard

 $\beta^{BN(P)}$ - Sensitivity of Earnings to Bad News

 $\beta^{GN(P)}$ - Sensitivity of Earnings to Good News

CAPM - Capital Asset Pricing Model

CoE - Cost of Equity

CEE - Capital Employed Efficiency

CE - Capital Employed

CIV - Calculated Intangible Value

CONS - Conservatism of Earnings

CFOs - Chief Executive Officers

DDM - Dividend Discount Model

DICM - Direct Intellectual Capital Methods

DPS - Dividend Per Share

E - Earnings of the Firm

EARN - Earnings before Extraordinary Items

ΔEARN - Earnings before Extraordinary Items in year t Minus Its Annual

Earnings in year t -1

EPS - Earnings Per Share

EPSHCE - Interaction of Earning Per Share and Human Capital Efficiency

EPSSCE - Interaction of Earning Per Share and Structural Capital

Efficiency

EPSICE - Interaction of Earning Per Share and Intellectual Capital

Efficiency

ERC - Earnings Response Coefficient

EVA - Economic Value Added

ERM - Enterprise Risk Management

FASB - Financial Accounting Standard Board

FCFE - Free Cash Flow to Equity

GDP - Gross Domestic Product

GLS - General Least Squares
GMM - Gordon Growth Model
GNP - Gross National Product

GAAP - General Accepted Accounting Principles

g - Growth Rate

HC - Human Capital

HCE - Human Capital Efficiency

IC - Intellectual Capital

ICE - Intellectual Capital Efficiency

IFRS - International Financial Reporting Standards

IPOs - Initial Public Offerings

ICAEW - Institute of Chartered Accountants in England and Wales

IAM - Intangible Asset Monitor

IT - Information Technology

KLCI - Kuala Lumpur Composite Index

KLSE - Kuala Lumpur Stock Exchange

KSE - Kuwait Stock Exchange

LCM - Lower-of-Cost-or-Market

MCM - Market Capitalization Methods

MASB - Malaysian Accounting Standards Board

MFRS - Malaysian Financial Reporting Standards

OECD - Organisation for Economic Co-operation and Development

OLS - Ordinary Least Squares

P - Price Per Share

PEG - Price-Earnings Growth

P/E - Price-Earnings Ratio

R - Return of Firm

r_{e, Re} - Rate of Return (Cost of Equity)

R_d - Cost of Debt

R_P - Cost of Preferred Stock

R&D - Research and Development Expenditures

R_f - Risk-Free of Return

R_m - Market Return

r - Pearson Correlation Coefficient

REL - Relevance of Earnings

RELHCE - Interaction of Relevance of Earnings and Human Capital

Efficiency

RELSCE - Interaction of Relevance of Earnings and Structural Capital

Efficiency

RELICE - Interaction of Relevance of Earnings and Intellectual Capital

Efficiency

RP - Risk Premia

ROA - Return on Assets

RIV - Residual Income Valuation

ROE - Return on Equity
SC - Structural Capital

SCE - Structural Capital Efficiency

SCM - Scorecard Methods

SEC - Securities Exchange Commission

SFAS - Statement of Financial Accounting Standards

VAIC - Value Added Intellectual Capital

VIF - Variance Inflation Factor

WACC - Weighted Average Cost of Capital

LIST OF APPENDICES

APPENDI	X TITLE	PAGE
A	Results of the Analysis to Test the Assumption of	307
	linearity of regressions: Residual Statistics	
В	List of Publications	314